

Received August 21, 2018, accepted September 17, 2018, date of publication October 3, 2018, date of current version November 8, 2018. Digital Object Identifier 10.1109/ACCESS.2018.2872686

Wireless Social Networks: A Survey of Recent **Advances, Applications and Challenges**

FURQAN JAMEEL¹, (Student Member, IEEE), SHURJEEL WYNE^{®1}, (Senior Member, IEEE), DUSHANTHA NALIN K. JAYAKODY^{®2}, (Senior Member, IEEE), GEORGES KADDOUM¹⁰³, (Member, IEEE), AND RICHARD O'KENNEDY⁴ ¹Department of Electrical Engineering, COMSATS University Islamabad, Islamabad 45550, Pakistan

²School of Computer Science and Robotics, National Research Tomsk Polytechnic University, 634034 Tomsk, Russia

³Department of Electrical Engineering, ETS University, Montreal, QC H3C 1K3, Canada ⁴Vice President Research, Development and Innovation, Qatar Foundation, Doha 5825, Qatar

Corresponding author: Dushantha Nalin K. Jayakody (nalin@tpu.ru)

This work was supported by National Research Tomsk Polytechnic University through the framework of Competitiveness Enhancement Program under Grant TPU CEP_IC_110.2017. The work of S. Wyne was supported by the EU-Funded Project ATOM-690750, approved under Grant H2020-MSCA-RISE-2015.

ABSTRACT With the ubiquitous use of smartphones and other connected pieces of equipment, the number of devices connected to the Internet is exponentially growing. This will test the efficiency of the envisioned 5G network architectures for data acquisition and its storage. It is a common observation that the communication between smart devices is typically influenced by their social relationship. This suggests that the theory of social networks can be leveraged to improve the quality of service for such communication links. In fact, the social networking concepts of centrality and community have been investigated for an efficient realization of novel wireless network architectures. This work provides a comprehensive introduction to social networks and reviews the recent literature on the application of social networks in wireless communications. The potential challenges in communication network design are also highlighted, for a successful implementation of social networking strategies. Finally, some future directions are discussed for the application of social networking strategies to emerging wireless technologies such as non-orthogonal multiple access and visible light communications.

INDEX TERMS 5G networks, centrality, community, social networking.

I. INTRODUCTION

A recent estimate reveals that on average, every third individual uses more than two devices connected to the internet [1]. The internet traffic involves the daily transfer of about 40 Exabytes of data [2]. This large data volume is commonly attributed to the widespread use of smart devices such as smartphones, tablet computers, smart TVs and smartwatches that are integrated with the Internet.

The need to share information has always been one of the key aspects of human survival. People meet with other individuals in their social networks to share their first-hand experiences and activities of common interest. This has led to the well-explored field of sociology, which is the study of human social behavior. Today, the ubiquitous Internet connectivity facilitates people to remain updated about the world around them. This Internet-based surge in social networking poses an important question: can we use strategies derived from sociology to improve the performance of future wireless communication systems? A straightforward reply is yes, because

the social networks we inhabit are related to the nature of the information that we consume [3]. In particular, our economic and information systems have become reliant on the structure of information such as people's preferences, interests, and hobbies [4]. Inspired by this phenomenon, many researchers have investigated the use of social networking features for improving the underlying communication technology, e.g., the proposal of context-aware wireless communication protocols [5]. Thus, the social network characteristics need to be studied for their judicious utilization in wireless communications.

Over the past decade, several authors have explored different aspects of social networks in wireless communications. Kayastha et al. [6] reviewed the application of social networking to mobile networks. They indicated that social networks can provide health-based services along with location-based and behavioral monitoring services. Later on, Zhu et al. [7] investigated the effects of social networking characteristics on the performance of delay tolerant wireless



FIGURE 1. Roadmap of social networking approaches for 5G and beyond. The applications of social awareness in wireless networks are shown together with some research challenges for its successful incorporation into the communication network design.

networks. The authors identified some positive contributions of the social characteristics such as assistance in message transmission, especially during node mobility, to improve the network's routing performance. The same authors also identified other negative behaviors, such as selfishness and misbehavior of nodes, which can compromise the network security. An overview of social-aware routing protocols from 2007 to 2012 was provided by Wei et al. [8]. Similarly, Singh et al. [9] performed a brief survey of network developments from the socio-5G perspective, whereas, Li et al. [10] established a new paradigm for analyzing the performance of device-to-device (D2D) communication by exploiting socially aware characteristics. They quantified the gains that can be obtained from social-aware D2D communications over the conventional D2D communications. For vehicular networks, Lequerica et al. [11] have reviewed various studies on social networking approaches to cope with the issues of routing and message transmission.

The proposed work is different from the aforementioned surveys in that we focus on providing detailed applications of social networks and the associated challenges for its practical implementation. A roadmap of social networking approaches for 5G communications is provided in Figure 1. Specifically, the major contributions of this work are listed as follows:

- A comprehensive discussion on the architecture of social networks and detailed classification of different social characteristics along with major vulnerabilities have been discussed.
- Case studies of various applications of social networking have been provided. Some of these applications include message forwarding, dynamic caching and security enhancement.

- Social networking solutions for 5G wireless networks including the Internet of thing (IoT) communications, cloud computing, vehicular communications, and the associated open issues are described.
- Detailed discussion on the applicability of social networking approaches on some broader areas of communications, such as non-orthogonal multiple access (NOMA), visible light communication (VLC) and smart grid communication, has also been provided.

The remainder of the paper is organized as follows. In Section 2 detailed architecture of social networking has been given along with a discussion on some social characteristics and their vulnerabilities. Section 3 provides potential applications of social networks. Section 4 provides details of the recent advances in the envisioned 5G technologies from the perspective of social networking. In Section 5, some of the open research challenges are presented. Then, Section 6 provides a discussion on some broader perspective of social communication. Finally, Section 7 concludes this work. In addition to this, a list of commonly used acronyms is provided in Table 1.

II. SOCIAL NETWORK ARCHITECTURE: AN OVERVIEW

Social awareness is the study of contextual information and enables communicating devices to formulate efficient protocols for their communication. As this is a relatively recent area of research, the design of the associated network architecture continues to be investigated in the literature. A simplified architecture consisting of six phases, i.e., sensing, analysis, knowledge, communication, cooperation, and coordination & collaboration, is illustrated in Figure 2 and is explained below.

At the physical layer, intelligent sensors are installed in the devices to obtain raw data from the environment [12]. At this stage, the devices are unaware regarding the existence of each other in the network. However, a set of devices may have a particular task and the collected data will be a representation of their similar interest. For instance, smart bulbs in a room will collect information regarding the room luminosity, whereas, a smart thermostat will collect the data on temperature and humidity in the room. Similarly, a smart entertainment system will collect the information related to users' experience and their viewing/ listening habits. The collected data from sensors is analyzed to obtain useful information by utilizing technologies like machine learning or data mining [13]. The information obtained from the environment can now be used for improving human-to-human, manto-device, and device-to-device interaction. At this level, the devices are somewhat aware of the situation and communication conditions.

After processing the obtained information, the devices are ready to plan on how to obtain knowledge regarding a given problem. Previous experiences can also be leveraged by the devices to acquire new information. After gaining a significant amount of knowledge about their surroundings, the social devices can now start communication with

TABLE 1. List of acronyms.

Acronym	Full form
ARIMA	Autoregressive integrated moving average
ATM	Adaptive trust management
BS	Base station
CBSTM-IoT	Context-based Social Trust model for the
0201111101	Internet of Things
CCRCN	Content-centric residential community net-
centert	work
CALA	Continuous action set learning automata
D2D	Device-to-Device
DONA	Data-Oriented Network Architecture
DHT	Distributed hash tables
DRU	Discovery resource unit
DSSoT	Dynamic social structure of things
FIB	Forwarding information base
FST	Firefly spanning tree
GSMA	Group special mobile association
HAP	High altitude platform
IaaS	Infrastructure as a Service
ICN	Information centric network
IoT	Internet of things
LBR	Low bit rate
LBR	Low on face Lookup based resolution system
LED	Light-emitting diode
LTE	Long-term evolution
MANETS	Mobile ad hoc networks
mmWave	Millimeter wave
NBR	Name-based routing
NDN	Named data networking
NDO	Named data objects
NEV	Network function virtualization
NetInf	Network of information
NOMA	Non-orthogonal multiple access
NRS	Name resolution system
OBD2	On Board Diagnostics 2
OFDMA	Orthogonal frequency division multiple ac-
OIDMIT	cess
ONSIDE	OpportuNistic Socially aware and Interest-
01.01012	based DissEmination
PaaS	Platform as a Service
pCoCe	Probabilistic control centrality
PoC	Proof-of-concept
PURSUIT	Publish-subscribe internet technology
REK	Reputation Experience Knowledge
ReViV	Rebroadcaster selection mechanism for
	Video streaming for VANET
RpR	Recommendations plus reputations
RSC	Route stability clustering
RSU	Road side unit
SaaS	Software as a Service
SIC	Successive interference cancellation
SIoT	Social IoT
SoCast	Social-aware video multiCast
SOR	Social on road
SPC	Sociological pattern clustering
SPRING	Social-based Privacy-preserving packet for-
	wardING
SPRINT	Social PRedIction-based routing in oppor-
	tunistic NeTworks
SUMO	Simulation of urban mobility
TACIoT	Trust-aware access control system for IoT
TRIAD	Translating relaying internet architecture in-
	tegrating active directories (TRIAD)
UAV	Unmanned aerial vehicle
VANET	Vehicular ad-hoc networks
VLC	Visible light communication
VPC	Virtual private community
Wi-Fi	Wireless fidelity
WIMAX	Worldwide interoperability for microwave
	access

other objects of interest in the network [14]. For instance, a smart lighting system can inform the smartphones in case of any anomalies in the electric power or malfunctioning of

the system. The ability to communicate with other objects in network open new opportunities to ensure cooperation between various devices. The exchange of information and resources and between devices happen at this level. Based on our previous example, a lighting system in a house may not be able to directly make contact with the smartphone a user who is far from home. Therefore, it can use the resources of wireless fidelity (Wi-Fi) to send its messages via the internet. Lastly, at coordination and collaboration stage, a social network harmonizes all the activities that need to be incorporated in order to work like a well-oiled machine. Later, these activities are given proper sequencing through collaboration among devices to perform effectively [15]. To achieve a common goal, devices use trusted social relationships which facilitate them to make proper decisions. Social-aware routing protocols and data dissemination schemes can be formulated here based on the information gained during the initial phases.

The above-mentioned functionalities can easily be incorporated in the latest devices which is one of the reasons for the dramatic popularity of social network theory in recent years. Presently, the devices are equipped with high-speed processors, large storage and wireless interfaces. With the help of short-range technology like Bluetooth and Wi-Fi, portable devices can now easily create ad-hoc networks to conveniently communicate with each other. Of late, research efforts to improve mobile communication have made great strides. Technologies like mobile sensing [16], [17], mobile ad-hoc networks (MANETs) [18], mobile computing [19] and mobile cloud [20] have surfaced which utilize opportunistic network approaches. The development of these fields, in parallel with the development of social networking, will immensely improve the performance of future networks.

Another important aspect of incorporating social behaviors into devices comes from the fact that most of the mobile devices are being used by humans. Smartphones are being carried by the people to their homes and offices. The mobility of these smartphones indicates the mobility pattern of the people. This mobility determines the properties of various social relationships as the people spend time with people whom they trust and have strong social ties. It is quite possible that people from the same community often contact each other for sharing information. This information can be extremely helpful for the devices as they can make decisions and design routing protocols for forwarding of data. Based on these social behaviors, the devices can also use prediction models to optimize their performance and save resources when no contact opportunity is available.

A. SOCIAL CHARACTERISTICS

Since most of the network entities are connected to each other in one way or another, therefore, the importance of social networks has risen exponentially [6]. Social characteristics not only define the relationship between devices but also illustrate the common interest and goals [21], [22]. We now discuss some of the key characteristics of social relationships,



FIGURE 2. The six design phases of a social network architecture. These phases are numbered in an increasing order of trust required between the communicating devices.



FIGURE 3. A typical mapping of physical infrastructure to a social network. The two access points of a cellular network, with their respective associated users, can be viewed as two social communities A and B. The set of users act as community members.

as shown in Figure 3, that are essential for the realization of the next-generation wireless network.

A social tie is a basic notion that indicates how closely two devices are related to each other. For humans, it is observed in activities and altruistic behavior of humans [24]. Similarly, in mobile networks, it is identified by how strong the two devices are connected. Based on these ties, a community is formulated between a group of people that share a common interest and demonstrate similar behaviors [25]. In the case of mobile networks, it may represent a cluster of devices located within a geographical proximity or having the same content. The communities are generally connected through a set of bridging nodes. A bridge is a connection between communities which is used to exchange information among adjunct communities. In general, every community has a pre-specified number of nodes for establishing a connection with the neighboring community.

Two nodes communicating frequently may have a significantly large contact history [26]. Thus, it can be intuitively concluded that these nodes have a strong social relationship. In contrast, the nodes that contact seldom to each other are considered to have a weak social connection. The social contacts can be characterized by multiple factors such as the contact frequency (which is the number of times contact was made between two or more nodes), contact interval (which is the reciprocal of contact frequency) and duration of contact (which is the average time during which the contact remained established). Another important characteristic is the social similarity of nodes. Social similarities can be found by using the contact books or records of the nodes in a network [27]. The nodes that most often discuss a particular issue or topic can be deemed as socially similar to each other. In other words, it can be defined as the closeness of two nodes that is determined from the intersection of their respective areas of interest. A trust value which is the expectation value of confidence that a node can put in the network. In simple words, it is the amount of trust that a node can put into the actions of a specific node. In addition, confidence value also relates to the trustworthiness which is the statistical probability of evaluated trust value. Both of these values are



FIGURE 4. Trustworthiness framework for determining the commitment and trust of devices [23].

calculated using a trustworthiness mechanism which is shown in Figure 4.

Centrality refers to the ability of devices to connect to other devices. A central user in the network can easily establish a connection with other network entities. One of the most common ways to measure centrality is by Freeman's degree [6].

The basic idea behind centrality is to gauge the effect that inter-node distance has on the communication between these nodes. Specifically, a node is more likely to communicate with its nearby neighbor than with a node located farther away. There are broadly three classes of centrality as shown in Figure 5. These are known as closeness centrality (which is described by the distance to reachable nodes), betweenness centrality (which is defined by the set of shortest paths from one node to the others), and degree centrality (which is the importance assigned to a node based on its ability to act as a bridge between different communities). The nodes may try to cooperate with each other, e.g., by relaying each other's data. Its examples can be found as dissemination of contact and relaying of data. A node in the network may be more willing to help a particular set of nodes than helping the others. This inclination towards a particular group can be because of the attached incentive in the form of allocation of some resource such as power or bandwidth, which would not be available otherwise.

B. SOCIAL VULNERABILITIES

Here, we provide some of the most discussed vulnerabilities in the literature on social networking. Due to the anonymity of nodes, it is quite possible that the malicious nodes provide false information or dishonest recommendations about themselves [28], [29] to increase the trust values. This is one of the easiest ways to spread wrong information about a node in the network which can be exploited in future communication. Moreover, a good node, if compromised, can become a malicious entity while an incompetent node can become a competent subject to different changes in the network. The on-off attack in social networking is indeed the exploitation of this behavior of nodes [30], [31]. It means that a malicious/ compromised node can act as a competent node for some time while it can act badly for other times. This strategy is usually adopted by nodes to stay undetected in the network while silently damaging the resources of the network.

Another commonly known attack is self-promotion attack in which a device provides bogus good recommendations regarding itself [32]. This results in building their good reputation among other devices. It also causes frequent selection of these devices for the provisioning of trusted services to other devices. However, these devices in actual provide low-quality services which affect the network performance. A node can act as good or bad in the time domain, but it can also show conflicting behavior in the user domain. Particularly, a malicious node can behave differently for different sets of users. This results in the conflict of opinions made by two sets of nodes whereby one set has a high recommendation value and the other set has a low recommendation value. In the Sybil attack, a node can take several IDs at the same time [33]–[35] which it can use to its advantage. More specifically, a node can use fake IDs to act as another node in the network. In this way, it can gain the trust of nodes that actually may not recommend it. Moreover, it can share the blame on other competent nodes resulting in damaging other nodes' social reputation.

Reputation lag in networks is the interval between generation of sensing report and determination of trust rating based on the mentioned report. During this time, an adversarial device can contribute towards sending false information regarding trust recommendations. Hence, this time lag can be misused by a device to provide corrupted services while quality services can be provided during sensing time to remain hidden. A malicious device can also launch an opportunistic service attack when it observes that its reputation in the network is falling. One of the reasons for this fall can be the provisioning of bad services by the devices. In this case, the adversarial device can start providing good services to the network for a brief period of time. When obtained a sufficient level of reputation, it can again start manipulation of data by colluding with other malicious devices [36]. In addition to this, a compromised node can exploit the relationship register to remove its previous bad record [35], [37]. If a node can easily register itself as a newcomer in the network then it can repeatedly exploit the system by exiting the network and returning a new node. This heavily damages the trust management flow of the network. Based on the above discussion, we now provide a detailed classification of social attacks in Table 2, with respect to non-cooperative or cooperative attacks and their direct or indirect impact on the network.

C. SUMMARY AND INSIGHTS

In this section, we have delineated some of the basic concepts of the social networking theory that are helpful to understand the remainder of this work. After showing a stepwise approach to design a social network of wireless devices, we have highlighted the fact that the level of trust, at each step, increases as devices move from the sensing phase to the collaboration phase. We have also identified the key characteristics of these social networks including social ties, social similarity and centrality. Figure 3 has shown the mapping of a typical physical infrastructure of wireless devices to a social network composed of different communities. Finally, we have highlighted some key vulnerabilities of social

Attack	Single	Group	Direct involvement	Indirect involve- ment	Non-cooperative	Mis-behaving	Compro-mised
Bad mouthing at- tack	√			\checkmark	\checkmark	\checkmark	
On-off attack	\checkmark		\checkmark		\checkmark	\checkmark	\checkmark
Self-promotion attack	√			V	V		
Conflicting behavior attack	√	√		\checkmark		\checkmark	\checkmark
Sybil attack	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	√	
Exploitation of reputation	~			\checkmark		\checkmark	\checkmark
Opportunistic service attack	√	√		\checkmark	\checkmark	\checkmark	
Newcomer attack	\checkmark		\checkmark	\checkmark	\checkmark	\checkmark	

TABLE 2. Classification of various attacks.



FIGURE 5. Classes of centrality and their closed-form expressions.

networks in terms of self-promotion attack, conflicting behavior attack, and Sybil attack. These vulnerabilities require research attention and long-term solutions for an optimal utilization of social networking approaches in 5G networks.

III. APPLICATIONS OF SOCIAL NETWORKING

We now discuss some of the recent advances in the application of social-aware networking to wireless communications, as shown in Figure 6. More specifically, we have categorized the social networking literature based on how the information extracted from a social relationship can be leveraged to improve the quality of service in a wireless link.

A. DYNAMIC CACHING USING SOCIAL NETWORKING

Efficient content delivery has gathered much attention in the recent years as it can significantly reduce the network traffic [38], [39]. Based on the recent research efforts, the content-caching technologies can be classified into the following two categories:

- 1) End Caching Caching is performed at end devices only.
- 2) Network Caching Caching is performed on the devices inside the network.

For content delivery networks, the network caching holds considerable importance and has been studied in detail over the past few years. However, the edge caching is relatively new and comparatively difficult to implement. In particular, recent developments show that the content can be cached at

assified into the at end devices to the problem und formed on the Bastug et al. [41] of backhaul and in su machine learning at to the problem und Bastug et al. [42].

highly function mobile edge devices [38]. Future networks are expected to be dynamic and flexible in a way that the end users will be able to receive not only content from the original server but also from the nearby mobile caching devices. As expected, this approach would reduce the end-to-end as well as content access delays and would also be helpful in reducing the traffic volume in the network. In this regard, some other recent developments are listed below:

1) RECENT ADVANCES ON SOCIAL-AWARE DYNAMIC CACHING

Owing to an increase in the demand of content in future networks, various strategies have been proposed to move the content near the end user [40]–[42]. Bastug *et al.* [40] explored the option of using small cells for caching, while in [41] and [42], the content itself was proactively downloaded by specific end users. These end users then distribute the content to other users via D2D communication. Bastug *et al.* [41] optimize the problem of load balancing in backhaul and in small cell connections. They also propose machine learning algorithms to heuristically find a solution to the problem under minimum iterations. In a similar work, Bastug *et al.* [42] use a prediction approach to find the most popular content within communities. The content is then stored at the base station (BS) of the small cells in a greedy approach.

To extend the above works, Siris *et al.* [43] introduced the aspect of mobility of users in the popularity prediction model. In particular, the proposed scheme determines whether an



FIGURE 6. Applications of social networking approaches. These applications consist of dynamic caching, resource allocation, link prediction, message forwarding, mobility prediction and security enhancement.

item should be saved at a particular BS of the small cell. The prediction model exploits the inter-arrival time of request generated within a community for a particular object. The authors proved that the performance under joint mobility and prediction model surpasses that of only mobility or only prediction. Due to the random nature of users' demands, it is generally difficult to model the content requests. In this regard, tools from stochastic processing can be used to model the uncertainties of content requests. With this intent, Tadrous and Eryilmaz [44] utilize probability distributions for modeling content requests uncertainties in proactive caching of content. Caching policies were also derived by the authors to show that lower bound can be achieved. Another technique to produce efficient caching was introduced in [45] for personalized content pricing schemes. An optimization problem was formulated and consequently, a sub-optimal solution based on an iterative algorithm was derived to separate the proactive download decision from price allocation strategy.

Huang and Cai [46] devised a caching mechanism by exploiting the social characteristics of the smart devices in the network. A welfare maximization scheme was formulated and solved by using the transformation method. The authors also demonstrated that both the BS utilization and network capacity can be significantly increased by exploiting the social characteristics of the intelligent network nodes. Wu *et al.* [47] designed a context-aware caching system, which employed a performance optimization routine based on the device capacity and real-time network conditions. Although the authors performed experiments in London City to verify their design, they asserted that their design was equally applicable to other major metropolitan cities like Shanghai, Paris, and New York City. Machado *et al.* [48] analyzed the spatio-temporal features of human mobility in New York City to investigate its impact on content dissemination. They also proposed a dynamic caching framework based on the analyzed spatio-temporal characteristics and social factors and verified their design through field trials.

2) CONCERNS AND LIMITATIONS ON DYNAMIC CACHING

Although edge caching appears to be a promising approach, yet the devices may not be completely willing to cache the data that they do not require. In other words, there can be a tendency of selfishness among different devices. Existing works assume the cooperative behavior of the devices, however, it may be possible that devices do not cooperate without any real incentive. Some incentive mechanisms have been proposed, still, the nature of incentive and its requirement may vary based on the type and size of the content.

Another concerning issue is the security of cached data. Proper mechanisms for security classification of cached content need to be defined. Moreover, there is a need to certify who would have the access to the cached content. All of these factors would require consideration the prior implementation of the social-aware content caching.

B. RESOURCE ALLOCATION IN SOCIAL NETWORKS

In order to meet the requirement of high data rates for new applications, wireless communication has been continuously evolving. Thus, for a reliable communication between two end users to incorporate increasing the number of users in the network, orthogonal multiple access techniques were proposed. Some relatively new systems like the worldwide interoperability for microwave access (WiMAX) and longterm evolution (LTE) systems use orthogonal frequency division multiple access (OFDMA). OFDMA divides a single bit streams into multiple streams, which are transmitted over orthogonal subcarriers. Although the OFDMA subcarriers improve the spectral efficiency, it may not be enough to cater to the demands of future networks. For an optimization of the resources, it is critical to have a clear view of the characteristics of the resources, the physical topology of the network and the status of entities in the network.

In addition, the resource allocation must be performed in a dynamic manner, allowing the resources to be released in case of a link or node failure. Moreover, the reconfiguration time of the system must also be taken into account to provide much-needed elasticity in resource demand. Keeping in view the importance of the resource allocation and its ability to impair the functionality of a wireless network, we now provide a discussion on the recent advances in social-aware resource allocation.

1) STATE-OF-THE-ART FOR SOCIAL-AWARE RESOURCE ALLOCATION

Resources can be efficiently managed by using social aspects of the network such as interactions, interests, and activities. Using this information, social utility functions and distributed game-theocratic approaches can be formulated. Semiari *et al.* [49] propose a context-aware framework to determine the strongly connected users in the network. The solution provided by the same authors has been found to be efficient when social similarities exist. Semiari *et al.* [50] dealt with the problem of joint allocation of resources for BS using millimeter and micrometer waves. Additionally, Namvar *et al.* [51] and Zhang Namvar *et al.* [52] performed resource allocation for small cell BS and for device offloading techniques, respectively.

Shah and Fapojuwo [53] investigated the impact of socialawareness in an OFDMA network. Two optimization problems were formulated to address the problem of resource allocation whereby devices directly communicate with BS during the first hop which can be extended to another hop in order to communicate with another device. The solution was compared with conventional resource allocation schemes to prove the usefulness of the proposed scheme. Li et al. [54] introduced a social community-based resource allocation technique by leveraging the properties of centrality and community. The proposed allocation scheme demonstrated better performance as compared to other contemporary schemes. However, the centralized nature of this scheme puts an excessive burden on BS and may introduce a single point of failure. Zhao et al. [55] maximized the overall throughput of the considered network by using a clustering approach. They selected the cluster head and cluster members by taking into account not only the geographical closeness between the nodes but also their social attributes. Subsequently, they used a matching algorithm employing a one-to-many bipartite graph for a fair allocation of the cellular channels. Their proposed scheme was shown to increase the throughput by 50% and 5% when compared with heuristic and stochastic algorithms, respectively.

In another related work, Zhao *et al.* [56] proposed a manyto-one matching algorithm with manageable complexity in which the user pairs and resource blocks interactively converged to a point of stability. Floréen *et al.* [57] showed that their proposed scheme outperforms the context unaware, oneto-one matching, and Gale-Shapley (GS)algorithms by 63%, 20%, and 11%, respectively. Zhang *et al.* [58] analyzed the results of an online interaction between users for improving the network performance. In [59], a resource allocation framework was developed for a media cloud. In particular, Tran *et al.* [60] used the backward induction method to obtain Stackelberg equilibrium. It was shown that each player in the game can achieve maximum revenue by exploiting the social features.

2) CONCERNS AND LIMITATIONS

Dynamic resource allocation has several drawbacks. For instance, most of the resource allocation based techniques consider perfect knowledge regarding the devices in the network. This knowledge is including that the channel state and distance are collected from the devices deployed in the network. By exploiting this information, it is possible that some malicious devices may gather more resources than they require. Hence, reducing the available resources for other needy devices. Future social network solutions need to jointly consider the resource allocation and trustworthiness of nodes for justified allocation of resources.

C. LINK PREDICTION AND SOCIAL NETWORKS

Link prediction is essential for characterizing the link-level performance of wireless networks. Link prediction works by detecting the missing or newly added links in the network. It can also help to identify the links that have a high probability of contact which is very beneficial for opportunistic networking [61]. Typically, the link is predicted using a link prediction score and this score is obtained using one out of the two methods stated below:

- 1) *Supervised Link Prediction:* In this approach, a classifier is constructed which predicts whether a link exists between two devices using previously learned characteristics of the network topology.
- 2) *Unsupervised Link Prediction:* This approach generates the link prediction score using real network characteristics like a number of neighboring devices and number of paths, etc.

As explained earlier, the supervision of network behavior can become tedious due to dynamicity and scalability of future wireless networks. Thus, we now focus on unsupervised link prediction, which is considered more suitable for wireless networks. As shown in Figure 7, there are two subcategories of unsupervised link prediction, i.e., neighborbased link prediction and path-based link prediction. For generating the link prediction score, the neighbor-based approach locates the neighbor of each node, thus using only local information. In contrast, global information is used for generating a score with the path-based link prediction method.

1) RECENT ADVANCES ON LINK PREDICTION SCHEMES

Recently, many studies have focused towards using social characteristics like ties, bridging, and community, to solve the common problems in wireless networks. Due to this reason, the problem of link prediction has gained significant research interest [62]-[64]. In this domain, one of the initial works was done by Valverde-Rebaza and de Andrade Lopes [65]. They combined community information (social behaviors and interests) with network topology to predict links for Twitter. It was unveiled that the proposed solution can efficiently predict the future links in asymmetric large-scale networks. Similarly, Li et al. [66] found that centrality approach can also be used to improve the link prediction accuracy. It is because most of the time, devices like to connect with similar devices and prefer a central node. Using maximal entropy random walk, they proposed a centrality based link prediction method.

Based on the above-mentioned works on link prediction, Liu *et al.* [67] used social ties between neighboring devices in order to find the node connection likelihood. They pointed out that among other social characteristics, social ties



FIGURE 7. Link prediction techniques based on node similarity score.

are more appropriate to improve link prediction accuracy. Qiu *et al.* [68] proposed a novel event-driven model to observe network processes in real time. These observations are then used to better predict the future link connections. Some other studies like [69] investigate complex approaches like homophily (togetherness) and social balance to find social patterns in the network. In another work, Moradabadi and Meybodi [70] developed a link prediction algorithm namely, continuous action set learning automata (CALA). Specifically, the authors try to develop a feedback environment and formulate link prediction as a noisy optimization problem. To solve this problem, different combinations of environment feedbacks are obtained at different periods which have shown to be helpful in learning the true value of the link.

Similarly, Gündüz-Öğüdücü and Çataltepe [71] provided a time series based forecasting approach for prediction of the link. This method works by examining the changes in past nodes similarities which is then used to predict recurring or new links. It was concluded that the proposed method improves the performance as compared to benchmark methods. Lebedev *et al.* [72] apply efficient top-k routing algorithms to solve the link prediction problem. The article compares the top-k shortest path results with the state-of-theart baseline methods and it was shown that the top-k similarity measure outperforms the classical Adamic/Adar and Jaccard measures.

2) CONCERNS AND LIMITATIONS

Using global and local information for link prediction has its disadvantages. In particular, the global information regarding paths cannot be available under all conditions. Moreover, for large-scale networks, it may introduce significant delays while simultaneously choking the resources of the network. On the other hand, local information requires less information exchange and has lower communication overhead. However, the local information based link prediction would underperform for the mobile networks.

Overall, the future efforts must be focused towards hybrid solutions. The social networks need to be equipped with decision-making algorithms allowing a smooth transition from neighborhood-based link prediction to path-based link prediction.

D. ROUTING AND MESSAGE FORWARDING WITH SOCIAL NETWORKING

Delay tolerant and opportunistic networks have recently gathered much research attention as they do not require the construction of new infrastructures [73]. Let us consider an opportunistic network where a source S wants to transfer message M to a receiver R. Since the direct path from S to R is missing, the message will have to be transferred through intermediate nodes. One of the easiest ways to transfer this message to R is through epidemic forwarding. In epidemic forwarding, each node that receives the message forwards it to the other connected nodes. Although it reduces the end-to-end message delays, yet the consumption of network resources increases significantly. Moreover, epidemic routing can cause congestion in a large scale network which may halt the functionality of the network.

It is evident from the above explanation that the intermediate node selection is important for efficient and reliable routing of messages. Therefore, the design of routing schemes that have the high probability of delivering the message to the destination has been extensively studied in the literature. We now provide a comprehensive explanation on some of the recent message forwarding techniques for social networks.

1) LITERATURE REVIEW OF ROUTING METHODS FOR SOCIAL NETWORKS

The social networking structure offers unique opportunities to exploit the network characteristics due to its implementation at three levels i.e. 1) individual 2) community 3) entire network [74]. Nodes in a particular community have more choices to communicate with each other than the nodes of different communities. One of the pioneering works was done by Hui and Crowcroft [75] to use the communitybased relationship for the routing of messages. Subsequently, the same authors proposed three algorithms in [4], namely, K-CLIQUE, SIMPLE, and MODULARITY. These were based on the contact graphs in a distributed community rather than utilizing a conventional centralized community-based approach. Bulut and Szymanski [76] proposed a method to strengthen the virtual link between devices to form a friendly community. The friendship community is a subset of the community that experience link quality better than the pre-specified threshold. Some other works consider the relationship between communication wherein intercommunity based forwarding decision mechanisms were proposed [77].

To liberate the devices to communicate independently, Daly and Haahr [14], [78] introduced community independent schemes. These schemes use the betweenness centrality to formulate a utility function which reduces the load on central devices. Moreover, few other studies consider embedding contextual information to route the messages [79]. The mobility of human beings has also been used to find the best path for routing the messages. For instance, Yuan et al. [80] use Markov chain technique to estimate the mobility patterns of people to estimate centrality and strength of social ties. The aforementioned protocols use unicast routing to forward the message to the destination. In contrast to these protocols, multicast routing can also be used to improve the routing performance [81]-[85]. However, one of the main concerns of these multicasting protocols is the frequent changes in the number of group members with time, as devices may join or leave the group independently [83].

2) CONCERNS AND LIMITATIONS

The majority of the recent studies uses unicast socially aware algorithms to improve the performance of networks (see e.g. [4], [14] and references therein). However, from a social perspective, only a handful of multicast algorithms have been developed [86]–[89]. It is therefore interesting to investigate how social-aware multicast routing protocols can be used to improve upon the performance of unicast protocols. Additionally, it is also interesting to see how social-aware any-cast algorithms can be used to improve the performance of wireless networks.

E. SOCIAL NETWORKING AND MOBILITY PREDICTION

One of the most important features of wireless nodes is that they can be deployed at any place without disrupting the normal operations of the existing network. Nodes can also join or leave a wireless network, thus changing the topology of the network. This type of change, which rarely occurs in the network, is called weak mobility. Wireless networks are generally good at handling weak mobility by frequently updating the information of their neighbors. However, nodes can only perceive a change in their surroundings at the start of their active period. This results in a considerable delay in packet transmission when the change occurs in the network topology. The performance worsens even more when a change is encountered in multi-hop networks. Still, the delay introduced by weak mobility can be tolerable since the modification rarely takes place in the network.

In contrast to weak mobility, frequent failure of nodes and their physical movement fall within the gambit of strong mobility. In some cases, nodes may be attached to a moving vehicle/ person or they can be moved by external forces like water or air [90]. A strong nodes mobility can cause some of the following major problems:

- 1) It deteriorates the quality of the wireless link, resulting in increasing the packet retransmissions.
- 2) The mobility of nodes can introduce significant packet delays due to frequent changes in the route.
- Due to the mobility of the nodes, the process of neighbor discovery can produce considerable communication overhead.
- 4) The mobility can increase the collision between packets, as the neighborhood information may lead to inconsistencies due to nodes joining/ leaving the network.

Overall, it is essential to predict the mobility patterns of the nodes using contextual/ learned information. In this regard, explanatory discussion on some of the recent studies for social networks is provided in the coming subsection.

1) MOBILITY PREDICTION FOR SOCIAL NETWORKING

In order to mitigate the scanning overhead in the future networks, Wanalertlak et al. [91] provide a mobility prediction tool for the efficient hand-off. Based on the previous information regarding users the authors make use of handoff tables which are fed into a prediction software namely auto-regressive integrated moving average (ARIMA). This strategy returns the likelihood of the user to hand-off to a nearby access point. The social platform called Foursquare has been used to leverage the information to predict the mobility pattern of the user. The authors collected the data for one month across different venues and build a timestamped data set to formulate linear regression and M5 tree. However, the authors do not provide specific applications of their proposed prediction model. Besides this, the mobility of users has been characterized for urban environments in [92]. The authors, in addition to social information, gain benefit from the physical information of the network to eventually formulate a probabilistic model on real-life data of mobility

TABLE 3. Summary of mobility prediction techniques.

Reference	Input Parameter	Technique	Prediction Efficiency
[98]	Previously visited location	Markov chain model	High
[99]	Previously visited location	Density joinable cluster & Markov chain model	95%
[100]	Time series of locations	Cross correlation	Directly proportional to the series length
[101]	Time information along with visited loca- tions	Viterbi decoding and supervised learning	High
[102]	Sequence of locations	Hierarchical triangular mesh and Markov chain model	Low
[103]	Sequence of regions	Association rule along with pattern extractions	Average
[104]	Node's online connections	OpportuNistic Socially-aware and Interest-based Dis- sEmination (ONSIDE) algorithm	Average
[105]	Node's social connections, contacts history,	Social PRedIction-based routing in opportunistic NeT- works (SPRINT) algorithm	High
[106]	Real time mobility traces of nodes,	Recent Weighted Average Method	High

traces of users. A brief summary of different mobility prediction techniques along with their prediction efficiency is presented in Table 3.

In a similar way, Bapierre et al. [93] predict mobility based on joint information from temporal (location of nodes) and social features (e.g., the location of friends with respect to the node). Through simulations, the same authors try to validate the accuracy of mobility patterns. The idea of social assistance among BSs was proposed by Yi et al. [94] to efficiently estimate the cellular network. Here, a social graph was created where the edge represents the spatial correlation of traffic between the BSs. In this way, the most critical BSs are identified to subsequently perform the prediction of traffic. It was found that prediction with fewer than 20% errors can be ensured using the proposed scheme. Tsiropoulos et al. [95] optimize the performance of downlink resource allocation by utilizing the popularity of end users. A sub-optimal solution was provided by heuristically solving the optimization problem which was then verified using synthetic data.

Ning *et al.* [96] proposed a cooperative service access system for vehicular networks and quality optimization method. Initially, the authors investigate and construct the social relationship between vehicles on the road. The authors then propose a trajectory-based time prediction algorithm to minimize the effect of the unstable network topology. The simulation results indicate that the proposed scheme is more effective than other baseline algorithms. Yuan *et al.* [97] studied the mobility patterns by incorporating anomaly detection, location prediction, and time-aware interdependency between devices. The experimental results demonstrate that the proposed algorithm outperforms the baseline schemes proposed in the previous literature.

2) CONCERNS AND LIMITATIONS

A few social-aware mobility prediction solutions have been proposed in the literature, however, they do not cover all types of scenarios. For instance, the mobility prediction model developed for vehicles on the highway may not be suitable for predicting mobility in a dense urban environment and vice versa. Moreover, future networks are also expected to incorporate high altitude platforms (HAPs) which would consist of several unmanned aerial vehicles (UAVs). The velocity of UAVs is higher than the vehicles on the road which warrants new mobility prediction schemes. Moreover, the mobility pattern of UAVs would be in three-dimensions, and therefore, two-dimensional mobility models could not be applied on UAV social networks. Future research works on the social network must focus on these aspects of mobility prediction solutions.

F. SECURITY ENHANCEMENT USING SOCIAL NETWORKING APPROACHES

Inevitably, the impact of the security in social relationships is quite significant. While most of the studies consider that nodes in the network are randomly deployed and grouped into the source and destination pairs [107]–[110], some studies have proposed social models for securing the large-scale networks [111]–[115]. The following subsection discusses the recent developments with respect to ensuring the security of social networks.

1) RECENT DEVELOPMENTS TO SECURE SOCIAL NETWORKS It is well-established that the social ties indicate the willing-

It is well-established that the social ties indicate the willingness of devices to share resources and information. In cooperative or relay assisted communication, it is commonly assumed that the intermediate relay is trusted [111], [112] which may not be a practical approach. It is because a relay can belong to another network or may be a compromised device. To partly address this issue, Vastardis and Yang [116] recommended using social trust degree for relay selection. Subsequently, few studies have used social ties to provide link security in cooperative networks [113]–[115]. More specifically, Tang *et al.* [113] analyzed secrecy outage probability of a cooperative network from a social perspective.

Similarly, Ryu *et al.* [114] proposed a cooperative transmission strategy. They considered that the relay can act as a potential eavesdropper based on the social trust degree mechanism. Huang *et al.* [115] provided a jamming mechanism and power allocation strategy based on the social characteristics of devices. To adopt a more realistic approach, they considered that the channel state information of devices in unavailable at the transmitter. These studies, even though, evaluate social ties on the basis of either relaying or jamming operations, yet a more dynamic scenario, where both relays and jammers cooperate exploiting social trust is largely missing in the literature.

Secrecy capacity analysis for large-scale networks was performed by Zheng *et al.* [117] by using social relationships. Particularly, they derive the secrecy rate and secrecy throughput based on the social ties between source and destination. Chen *et al.* [118] exploit the social layer information for jamming the transmission of eavesdroppers. They leverage the social reciprocity along with social trust to improve the secrecy performance of the network. In order to extend their previous work [119] for cooperative networks, Wang *et al.* [120] used contact duration to propose admission policy in the network. Ergodic secrecy rate using these social characteristics was also evaluated without the knowledge of CSI.

2) CONCERNS AND LIMITATIONS

Establishing secure end-to-end communications between two legitimate devices is one of the most critical issues in wireless communications. Some factors contributing to the unsafe transmission of messages are an uncoordinated access of the medium, a centralized routing mechanism, and unreliability of neighboring devices. Although the social perspective in communications can strengthen the authentication aspect of link security, it has its inherent vulnerabilities, as discussed in the previous sections. For example, the information collected for employing trustworthiness can be exploited by a malicious user to obtain access to the data on other devices. Providing security, in this regard, is significantly difficult due to the availability of global state information in the network. Which device should have access to this information and what degree of access should be provided to these supervisory entities must be determined in order to materialize the social networking potentials.

G. SUMMARY AND INSIGHTS

This section has identified some application scenarios of social networking concepts in wireless communications. Specifically, social networking approaches can be used for dynamic caching, resource allocation, link prediction, mobility estimation, message forwarding, and security enhancements. Resource allocation and dynamic caching applications have been largely focused on optimizing the utility function under a context-aware framework. On the contrary, the link/ mobility prediction and message forwarding applications of social networking mainly rely on eventdriven models. Social networking approaches have also been reported to improve the security of wireless devices, but these studies are limited to a specific layer only. The proposed security solutions would require considerable refinements to make full use of the cross-layered architecture of wireless networks.

 TABLE 4. Some variations of the 5G vision, as proposed by different stake-holders [123].

Key Players	5G Vision
	- Universal filtered multi-carrier
5GNOW	- Unified frame structure
	- Abandon synchronism and orthogonality
5G Forum	- Intertwining heterogeneous networks
30 Porum	- Commercialization by 2020
	- Personal mobile Internet, D2D
5G Training	 Architecture and key technologies
	- Disruptive technology direction
	- Scalable and sustainable
5GPP, METIS	- Multi-tenancy
	- Softwarization of 5G
	- Extensive cloud computing
Samsung Electronics	- Enhanced multimedia experience
	- Internet of things
	- Sufficiently accurate channel models
NOKIA Solution Networks	- Augmented reality and tactile Internet
	- Heterogeneous deployment
DOCOMO	- Everything connected wirelessly
DOCOMO	- Extensive and enriched content
	- Network deployment scenario
Huawei	- Diverse application and services
	 Massive capacity and connectivity
	- Improved user experience
Qualcomm	- Connecting industries and devices
	- Enabling novel services
Ericscon	- Affordable and sustainable
Enesson	- Networked society

IV. POTENTIAL SOCIAL NETWORKING SOLUTIONS FOR 5G COMMUNICATIONS

Social networking approaches have the potential to exploit the social characteristics of networks to improve their performance. Lately, the research community has focused its attention towards using the social aspects of wireless networks. Thus, in this section, we discuss some recent developments in key enabling technologies of the 5G communication, from the perspective of social networking. Let us first understand the requirements of 5G communications as it will help us in analyzing some recent studies.

A. 5G-ENABLED WIRELESS COMMUNICATIONS

The wireless communication networks have been significantly improved over the past decade due to focused research efforts. The combined effect of increased scalability, hyperconnectivity, and application-specific requirements are helpful in triggering the next evolution of wireless networks (i.e. the 5G networks) [121]. It is also envisaged that 5G technology will bring the magnitude of the decrease in energy consumption and round-trip latency while increasing the data rates, connectivity, and coverage. As shown in Table 4, the key vision of 5G is different for top market players, which makes it difficult to shape standards for 5G technologies. Still, the standardization efforts are more active today than before and the first standard for 5G is envisioned to get mature by 2020. Group special mobile association (GSMA), along with its partners from industry and academia, is working on blending several research activities to meet the major requirements of 5G communications [122]. These requirements are listed below:

	1105	Druge	Centrality
ncounter patterns & peer discov-	Beacon rate adjustment	-	Communication demands
ry			& proactive beacons
community interests & community	-	Inter-community demands	Bottleneck detection &
ensity			cellular preferential
communication demands &	Security and privacy & communi-	Dissemination based bot-	Resource based bottleneck
ommunity-oriented sharing	cation demands	tleneck prediction	prediction
Distributed coordination & re-	Spectrum allocation & relay selec-	-	-
ource partition	tion		
	counter patterns & peer discov- mmunity interests & community nsity mmunication demands & mmunity-oriented sharing stributed coordination & re- irce partition	counter patterns & peer discov- // Beacon rate adjustment // - nsity - mmunity interests & community - nsity - mmunication demands & mmunity-oriented sharing Security and privacy & communi- cation demands stributed coordination & re- irce partition Spectrum allocation & relay selec- tion	counter patterns & peer discov- Beacon rate adjustment - // - Inter-community demands nsity - Inter-community demands mmunication demands & Security and privacy & communi- cation demands Dissemination based bot- tleneck prediction stributed coordination & re- tion Spectrum allocation & relay selec- tion -

TABLE 5. Application of social awareness to D2D communications [10].

KEY REQUIREMENTS OF 5G COMMUNICATIONS

The future networks are going to revolutionize our lives by leveraging large-scale connectivity of heterogeneous and smart devices. To materialize this vision, following key requirements have been established for 5G systems:

- *High data rates:* In order to provide somewhat acceptable performance for future applications (e.g., virtual reality, augmented reality, video streaming, etc.), the 5G technologies would have to support data rates up to 10 Gbps.
- *High scalability:* To support fronthaul network decomposition through network function virtualization (NFV), high scalability of networks is essential.
- *Low latency* Future applications like video games, tactile Internet and sophisticated car alert systems would require lower latency which cannot be supported through existing infrastructure. Therefore, 5G must support 10 times less round-trip time as compared to existing LTE systems.
- *Higher reliability:* 5G communications must be able to provide uninterrupted service to a massive number of devices.
- *Perceived availability of network:* The network coverage needs to be available for 99.9% of the time so that the network is available anywhere, anyplace and anytime [122].
- *Minimization of security threats:* The massive connectivity of devices may have catastrophic effects if a malicious user executes network attacks. It is, therefore, needed that 5G communications incorporate effective and light-weight security protocols to protect user privacy and to avoid network failures.
- *Increased battery life:* Due to rapid depletion of energy sources, green technology would be an essential part of future communications networks. Hence, a 5G system must support energy efficient methods and standard bodies need to incorporate this feature while making policies [122].

Based on the above discussion, it can be concluded that the exact concept of 5G communication is still vague. However, the 5G requirements can act as potential guidelines towards the realization of 5G systems. In this context, social networks can play a pivotal role in optimizing the performance of large-scale 5G networks. In light of this vision, we now analyze some recent studies on wireless social networks.

B. SOCIAL NETWORKING IN D2D COMMUNICATIONS

With the densification of cellular networks, the users in close proximity would be able to communicate directly without any involvement of BS. This communication approach between devices is called direct D2D communications. Due to close proximity, the social networking approaches can be readily applied to D2D networks. For instance, community-based techniques can be used for optimal allocation of resource and interference management in the network. Besides this, a summary of other qualitative attributes and design aspects of social networking enabled D2D communication are given in Table 5.

1) SOCIAL-AWARE RESOURCE ALLOCATION FOR D2D COMMUNICATIONS

Recent studies in D2D communications have revealed that social behaviors of human beings can be used to improve the performance of D2D networks [124]. Zhao *et al.* [125] provided a solution for optimal resource allocation by using utility games within a social group. Orsino *et al.* [88] and Zhang *et al.* [89] investigated cluster based formulation for content dissemination and for channel sharing under various duplexing schemes, respectively. For energy efficient multicasting environment, the heuristic algorithm approach achieves a better throughput per unit energy consumption [87].

A two-step game model was provided Wang *et al.* [126]. Specifically, the authors noticed that D2D pairs can reuse the channels of cellular communities thereby resulting in optimal resource allocation. A firefly spanning tree (FST) based algorithm was provided by Chao *et al.* [127] which used received signal strength to formulate the tree. By adopting a similar methodology, Tang *et al.* [128] presented an uplink signal detection scheme. However, one drawback of this strategy is that the privacy of users was neglected while discovering neighbors.

2) INTERFERENCE MANAGEMENT USING SOCIAL-AWARE D2D COMMUNICATIONS

In some cases, the communication overhead for device discovery becomes very large due to dense deployment of devices. To address this issue, social networking based adaptive approaches can be used for discovering devices in nearby proximity. Specifically, Zhang *et al.* [52] vary the probing rate of device discovery based on the information of the social domain. While in [129], the wake-up schedule of devices is changed according to the probabilistically evaluated contact time of users. Thus, the lifetime of devices can be increased by appropriately adjusting the accuracy of device discovery. Community-based attributes can be exploited in massive D2D networks to ensure reliable transmission of data within the community and across different communities. With this motivation, Hasan and Hossain [130] used community-based attributes to configure the transmission power of devices.

3) TRUST-BASED SECURE COMMUNICATION IN D2D NETWORKS

Trust of each node is calculated with respect to particular metrics such as trust in the fast delivery of a service or the trust to deliver high quality or either trust for using as a relay. Trust is calculated by taking into consideration experience from direct interaction, knowledge from other devices and timing information. Each node maintains a trust table of all its neighbors and selects its relay node based on updated trust values in the device reputation table. Trust in a social network can be further divided into two classes; user based and device based social features [131]. Based on this observation, Ometov et al. [131] propose a weighted function that considers both classes of social features. The function's behavior is dependent on the way the weights are assigned. Higher weights can be assigned to either user-based social metrics or device based metrics depending on the requirements of a particular application.

Social relationships and trust are exploited by Social-aware video multiCast (SoCast) [132] to propose a social-aware video multicast system for D2D communications. SoCast exploits the social trust and social reciprocity parameters to stimulate cooperation between devices. Users can form groups to obtain missing packets from other clients and restore incomplete video frames, thus, improving overall video quality and user satisfaction. However, the element of trust was missing when groups are formed in [132]. Since social network approaches mainly rely on the trustworthiness of nodes, Mishra and Pandey [133] proposed a more sophisticated trust based relay selection scheme.

A privacy-preserving spatiotemporal scheme is proposed in [134]. Spatiotemporal matching is based on the location dependent information of D2D devices. Spatiotemporal profile of each device is maintained by keeping track of devices' whereabouts. Spatiotemporal matching of profiles of two devices can be used to determine the level of trust the devices have on each other. Devices with similar spatiotemporal profiles are more likely to stay within the transmission range longer. Liu et al. [135] investigated socialaware content caching mechanism in order to reduce downlink latency. By exploiting social characteristics, the authors selected important nodes in the D2D underlay cellular network to allocate files. Based on their previous work [135], the same authors proposed many-to-one and many-to-many file allocation strategies in [136]. The authors proved that the proposed scheme outperforms the benchmark schemes by reducing the downloading latency of the D2D-enabled cellular networks.

C. SOCIAL NETWORKING ENABLED VEHICULAR NETWORKS

In the present era, an increasing number of people are using private vehicles and cars on a daily basis. This increased use of cars has resulted in an exponential rise in fatalities due to road accidents. The danger of these vehicles have been recognized by modern societies and several solutions have been proposed to minimize it. In this regard, wireless communications provide a potential solution in the form of safe and reliable vehicular networks. It allows vehicles to share information on the road which may include travelerrelated information, safety information to minimize the risk of accidents, and post-accident information for traffic jams and for accident investigation. This emerging field of communication has attracted the attention of researchers from all the domains including social networking. Some of the most prominent applications of the social network are appended below:

1) APPLICATIONS OF SOCIAL NETWORKING IN VEHICULAR COMMUNICATIONS

Network applications combining social networking and human factors can be naturally used for vehicular networks. A socialized vehicular network has common preferences, interests, and objectives. For instance, a group of people heading towards a football stadium to watch the same game can be expected to have the same interests which affect the vehicular connectivity [137]. Of late, mobile applications such as RoadSpeak and NaviTweet have been introduced to address drivers preferences [138], [139]. These applications are mainly devoted to providing infotainment and safetyrelated services. Another exciting aspect of incorporating social assistance in vehicular networks is that it can be used to reduce the traffic jamming conditions by predicting social behaviors of people [140]. In addition to this, people sharing same interests can discuss topics of mutual benefit [141] and form a community. Different communities can connect via intermediate helper vehicles which also act as bridges, as shown in Figure 8.

2) INCORPORATING SOCIAL BEHAVIORS IN VEHICULAR NETWORKS

Li *et al.* [27] proposed an efficient data dissemination scheme based on social similarity and local activity. However, two major concerns, as noted by Gillani *et al.* [142], in the social design of vehicular networks are (1) efficient integration of sensors with other technologies and (2) privacy of data. In this context, Lu *et al.* [413] proposed Social-based PRivacypreserving packet forwardING (SPRING) protocol which exploits the efficient deployment of a roadside unit (RSU) at social points of intersection. However, the establishment of trust among drivers is still a critical aspect of vehicular networks and principles of admissions in a social group need



FIGURE 8. An illustration of social-aware vehicular networks. The figure shows four communities of vehicles: Community A, Community B, Community C, and Community D. Different communities of vehicles communicate with each other through intermediate vehicles called bridges. The main challenge is related to frequent changes in the number of community members due to the mobility of vehicles.

to be clearly defined [144]. Normally, the following factors should be considered:

- Formation of a social group.
- Evaluation of trust and its management [145], [146].
- Establishment of decentralized architecture.
- Flexibility in data exchange and its integrity [147].

Some other vehicular technologies focus on network analysis to ensure web-based portal to register the interested users [148]. Particularly, the importance of a node in the network can be evaluated by exploiting centrality metrics. This metric is then used to estimate social ties between the vehicles in order to register the incoming ones [148].

3) PACKET FORWARDING IN VEHICULAR NETWORKS USING SOCIAL-AWARENESS

Bradai and Ahmed [149] proposed a novel mechanism to provide efficient video streaming to the users based on dissemination capacity. Cunha *et al.* [150] proposed a new solution to select the best vehicle for rebroadcasting the messages while Stagkopoulou *et al.* [151] introduce a new metric namely probabilistic control centrality (pCoCe) to select the best vehicle for forwarding the messages. Maglaras and Katsaros [152] introduce new techniques i.e. route stability clustering (RSC) and sociological pattern clustering (SPC) to estimate the tendency of vehicles to take a particular route. Among other clustering approaches, Vodopivec *et al.* [153] performed a review of cluster-based vehicular networks. Other noteworthy context-aware social

D. SOCIAL NETWORKING AIDED CLOUD COMPUTING AND BIG DATA

With the rise of social media, the amount of data captured by organizations has been increased manifolds. This data can be collected in either unstructured or structured format. The creation of data (hereby referred as big data) is occurring at a rapid pace and this trend has received worldwide recognition [161]. Big data is quickly transforming business, finance, engineering, science, healthcare, and ultimately, the entire society. One of the major problems that practitioners and researchers are facing relates to slow development in cloud computing platforms for the analysis of data. One of the main advantages of social networking approaches is that they can be applied to both big data and cloud computing.

The cloud computing has several advantages like scalable data storage, integration of data service, parallel processing, virtualized resources, and security. Resultantly, various applications have been developed that leveraging the cloud platforms to efficiently manage the data for user access [162]. Despite some recent efforts, very few works exist in the literature where social networking approaches are used for cloud computing and big data analysis. In the following subsections, we discuss these recent works that apply social networking for scalable data management in cloud computing and for the analysis of big data.

1) SOCIAL NETWORKING FOR CLOUD COMPUTING

Scalable data management has always been one of the key motivations of the research community. In this context, cloud computing has proven to be an extremely useful paradigm. Cloud computing has mainly three paradigms namely, Software as a Service (SaaS), Platform as a Service (PaaS), Infrastructure as a Service (IaaS) [163]. Some of the attractive features of cloud computing are a low investment, elasticity, and transfer of risk. It can provide novel platforms by enabling distributed and ubiquitous availability of data.

The concept of cloud computing has generated a tremendous proliferation of data-driven applications. To manage this distributed data has been a key motivation of the database researchers for the last three decades. Much work has been put into ad-hoc and update-intensive workloads to improve the scalability of networks. However, these distributed systems were not successful due to the commercialization of parallel databases [164] which significantly grew in numbers [165]. Rapid variations in the data access patterns of applications need to be carefully evaluated and, therefore, the design of a single perfect data management system for cloud platforms is still an open issue.

Chard *et al.* [166] formulated a social cloud environment by combining the concepts of volunteer computing, cloud computing, and social networking. They proposed a

TABLE 6. Overview of social protocols for vehicular networks.

Reference	Challenge	Solution	Network Scenario
[143]	Formulation of incentive mechanism for security and privacy	Packet forwarding protocol SPRING	RSU at highly social inter- active points i.e. V2I
[156]	Efficient treatment of data to ensure privacy of users	Packet forwarding protocol SPF	Social relay deployment for V2I communication
[157]	Trust and incentive-based user involve- ment	Packet forwarding protocol EVSE	V2I
[158]	Paradigm shift from centralized to dis- tributed network architecture	Modeling using SocialDrive	V2V
[149]	Reliable dissemination of messages	Packet (video) forwarding protocol Rebroadcaster selection mechanism for Video streaming for vehicular ad-hoc net- works (VANET) (ReViV) based on location and centrality metric	V2V
[159]	Treatment of data for effective dissem- ination of messages	Next hop forwarder protocol SCR based on social relations	V2V
[160]	Message dissemination and treatment of data	Packet forwarding protocol BEEINFO	V2V
[152]	Message dissemination	Packet forwarding protocol SPC/RSC	V2V

 TABLE 7. Summary of social approaches in cloud computing and big data.

Reference	Challenge	Contribution
[184]	Presented the idea of combining Big data along with	Analyzed the complexity of knowledge extraction when social networks
	social networking	are used to address Big data challenges
[185]	Improving data credibility and privacy	Provided practically executable network design and highlighted its appli-
		cations in healthcare
[186]	Integration of Big data and social IoT	Social graphs approach for modeling of connected nodes
[187]	Review of the social perspective of Big data	Provided challenges and research issue for social big data
[188]	Introduced the concept of joint utilization of social	Proposed RFTRS scheme while using reputation and topological informa-
	networking and cloud computing	tion of nodes
[189]	Service availability from the cloud for ad hoc net-	Presented the techniques to improve service availability whereby improving
	works	the efficiency of data forwarding
[190]	Providing mobile services using cloud computing	Proposed CloudMoV using IaaS and PaaS cloud and executed on Amazon
		EC2 and Google app engine
[191]	Efficient video streaming and cloud storage	Using CloudMoV proposed techniques to improve scalability and avail-
		ability of data and pre-fetching intelligence
[192]	Allocation of tasks of big data mobile networks	Proposed trust based game-theoretic incentive scheme for task allocation
[193]	Reducing the workload of content-centric networks	Proposed social utility maximization game and find socially aware Nash
		Equilibrium point
[194]	Performance evaluation in terms of revenue and	Developed a practical cloud system with the help of the CometCloud
	reputation	system

method where storage resources can be shared among friends. This model overcomes the restriction of sharing data among friends. However, it also assumes that the trust exists between users; which may jeopardize the security of data [146], [167], [168]. Of late, some incentive-based techniques have also been surfaced to promote the concept of social cloud computing [169]–[172]. This mechanism encourages devices to share their resource for providing data services to other devices. Bidoki and Kargar [170] proposed bee colony algorithm for sharing tourism-related data in the social cloud. Using an incentive approach, Chard and Caton [172] provide a crowd-sourcing infrastructure for social cloud network.

2) USING SOCIAL NETWORK FOR BIG DATA

Big data is the data analysis technique to support rapid analysis, storage and capturing of data. The data sources include numerous types of databases such as sensor data, mobile devices data and social media, etc [173]. In addition, the data need not be structured, rather the unstructured data that has no standard form can also be used. A typical data storage infrastructure for big data is in the form of clustered network attached storage [174]. Although the price of storage has significantly declined over the past years, it may still pose difficulties for small or medium-size businesses. However, despite recent advances, only a limited number of studies exist in the literature of big data that make use of the social characteristics of devices.

A brief summary of a few recent studies that investigate various aspects of cloud computing and big data from the social perspective is provided in Table 7.

E. SOCIAL NETWORKING ENABLED INFORMATION CENTRIC NETWORKS (ICNS)

Presently, internet communication is purely based on addressable hosts i.e. IPv4/IPv6 and these are rapidly exhausting leading to adopt the latest concept of naming content or data namely named data objects (NDO). This will, on one hand, eradicate the shortcomings of IP addresses but on the other, it will also be more susceptible to session termination, perturbations and other drastic anomalies in network operations. ICN vision was primarily initiated by translating relaying internet architecture integrating active directories (TRIAD) [175] and has been progressively researched by various interested exploring groups namely data-oriented network architecture (DONA), named data networking (NDN), publish-subscribe internet technology (PURSUIT) and a network of information (NetInf) [176].

ICN architectures, generally, have an edge repository cache technique spontaneous within the network. As opposed to the contemporary internet which enables the senders to be in-charge of data transmissions, the content in ICN is receiver-driven technically. It consists mainly of two phases: first, the discovery is triggered by the consumer to find the content or its duplication, and the second is taking it back to the concerned consumer. There are two methods of content discovery: name-based routing (NBR) and lookup based resolution system (LRS) [177]. In the NBR, the consumer sends a content request packet (i.e., the so-called interest), hop-byhop relaying by the packet forwarding nodes by looking up a name match into their forwarding information base (FIB) [178]. Whereas, with LRS, the content request corresponds to a resolution system, whose implementation is different in all ICN architectures. For instance, scalable and adaptive internet solutions (SAIL) protocol architecture defines a distributed name resolution system (NRS) that is based on hierarchically distributed hash tables (DHT). SAIL has been implemented as a hierarchical network and collects the publish messages and guides the layout (topology) manager for instilling compact path delivery.

1) CONTENT DISTRIBUTION TECHNIQUES USING SOCIAL-AWARENESS

Social networking approaches are a natural fit for the ICN and provide considerable improvements over the conventional TCP/IP-based networks. However, the work on the social perspective of ICN is very limited. Fleury [179] analyzed the interacting users to estimate the strength of the social relationship between them. Centrality measures such as PageRank [180] and Eigenvector [181] have also been used to improve the user interactions. Another important work is done by Bernardini *et al.* [182] who proposed socially aware caching for the ICN network. Mathieu *et al.* [183] noted that a content-centric network can employ short messaging services by introducing a fraction of complexity in existing networks.

In order to accommodate the demands of users, content delivery networks can exploit the social characteristics. In fact, some studies have proposed to use social characteristic evaluation models to further optimize the distribution of content on the Internet [195], [196]. For instance, Agarwal and Agarwal [195] analyzed Twitter's trending policies using an auto-regressive prediction method. Focusing on the geographic aspect of content-based networks, Scellato *et al.* [196] provided an efficient caching mechanism to store the most popular content on content delivery networks. Motivated by this, Nazir *et al.* [197] proposed a content delivery mechanism to extract time-based information and to estimate the contact period. Besides, distributed models, a central energy consumption model for content-centric networks can be used to make energy efficient caching decisions [198]. As a result, Fang *et al.* [198] proposed a gametheoretic strategy for making caching decision. A socially optimal configuration was found by deriving the Nash equilibria.

2) SECURING ICN USING SOCIAL NETWORKING

Kim *et al.* [199] introduced the concept of the virtual private community (VPC). VPC is an ICN based platform which facilitates users to communicate with friends and family without the need for a central server. Moreover, VPC does not require a centralized authentication server for verification of signatures. Kim and Lee [200] provided a proof-of-concept (PoC) where a selected group of members can download a list of content. Based on [199] and [200], Garroppo *et al.* [201] evaluated energy efficiency of a content-centric residential community network (CCRCN). To determine the energy footprint of the network, the same authors used an energy model. The energy model was constituted by the contents from the Internet, the power of gateways and the amount of exchanged content between the gateways.

F. SOCIAL NETWORKING AND IoT

The IoT has shifted the paradigm of device communication towards pervasiveness and heterogeneity, where a large number of connected devices generate information for particular applications [202], [203]. However, designing unique protocols to meet the demands of such a large number of devices is no simple task. The exchange of information needs to be efficient enough such that everyday objects can communicate/ interact using the minimum assistance of human beings. This has led to the introduction of the concept of virtual objects [204] that augments the abilities of real-world physical objects. As per [205], three phases can be identified enable physical-virtual world interaction:

- Information collection phase.
- Information transmission phase.
- Information processing phase.

1) ROLE OF SOCIAL NETWORKING IN IOT

Khan *et al.* [206] define SIoT as "A Social Collaborative Internet of Things is a new paradigm that has strong ties with Social Internet of Things and is defined as a platform of IoT where smart objects work together socially through recursive interactions of knowledge by establishing social relationships with their surroundings smart objects aiming to achieve common/shared goals in order to benefit humans". Based on this idea, Colom *et al.* [207] proposed a framework for load computing and devised a scheduling mechanism for allocating tasks to IoT devices. The scheduling is performed after a given device has computed the load of itself and its neighboring devices. Once the scheduling has been performed, the tasks are distributed among available devices.

Reference	Challenge	Contribution	Network Condi- tion
[215]	SIoT link selection problem	Performance analysis of social network of objects and adjustment of threshold for different numbers of users	Static
[216]	Performance analysis for cognitive Social IoT	Quantification of social relations for different characteristics of mobile nodes and managing irregularities of these social traits	Static
[217]	Modeling social network of objects and performance evaluation of web of things	Management of social network of objects with minimum involvement of humans	Static
[218]	Analysis of complex dynamics of SIoT devices	Explores dynamics and complexities of Social IoT and explore its application in disaster management scenario	Static
[140]	Implementation of IoT devices	Discussion on intelligent transportation systems station architecture functionalities	Mobile
[219]	Connection time prediction in SIoT	Proposed social on the road (SOR) for the case when number of users in the network changes and connectivity is irregular	Mobile
[220]	Architecture for SIoT	Performed simulations with Open Street Map, on board diagnostics 2 (OBD2) and simulation of urban mobility (SUMO) trace for various emergency applications	Mobile
[221]	Efficient resource utilization	Proposed light-weight query mechanism with the help of "Fission Computing"	Static

TABLE 8. Overview of recent developments in Social aware IoT.

Kasnesis *et al.* [208] designed two ontologies to handle the social relationship of devices followed by investigating use case scenario of social IoT. Hussein *et al.* [209] provided a social structure namely dynamic social structure of things (DSSoT) to evaluate the devices' short-term needs based on their situation. The algorithm consists of two phases (1) identification phase (2) goal detection phase. The performance of the system was evaluated in terms of complexity and data growth rate. A list of recent development with respect to efficient forwarding of messages in SIoT is provided in Table 8.

2) TRUST-BASED TECHNIQUES FOR SECURING IoT

As one of the aims of IoT is to autonomously make decisions for different services and applications, therefore, the importance of trust for SIoT cannot be neglected [210], [211]. A trusted platform can maximize the probability of the transfer of information and minimize the risk of eavesdropping. To build a trusted platform for SIoT, various factors need to be incorporated e.g. knowledge, experience, and reputation. Some other protocols for trust management in SIoT are adaptive trust management (ATM), recommendations plus reputations (RpR), trust-aware access control systems for IoT (TACIoT), context-based social trust model for the internet of things (CBSTM-IoT) [32], [212]–[214].

G. SUMMARY AND INSIGHTS

This section has provided the recent developments in 5G communications with respect to social networking. The introduction of social characteristics in 5G communications has completely altered the perspective of wireless communications. At present, studies are being conducted to improve the performance of D2D communications, vehicular communications, ICN & cloud computing networks, and IoT. For D2D communications, social characteristics like social ties can be used relay selection and beacon rate adjustment. Similarly, social centrality can be used BS for allocation of resources among D2D and cellular users. In the domain

of social-aware vehicular communications, solutions like SPRING and ReViV have been proposed to improve security and message dissemination, respectively. Centrality measures like Eigenvector and Pagerank have also been proposed for social-aware caching in ICN. Many investigations of social networking enabled IoT exist in the literature and proposals of fission computing and SOR have been given for static and mobile IoT, respectively. These studies indicate that social networking can be a promising approach for future 5G communications.

V. LESSONS LEARNED AND OPEN ISSUES

In view of this discussion, we now summarize some key challenges in the implementation of social perspective in existing and future wireless networks.

A. TRADEOFF BETWEEN COMPLEXITY AND PERFORMANCE

It is obvious that social-aware protocols require additional complexity and functionality in devices to perform as a community rather than as independent entities. Moreover, it has been proven that protocols which utilize a hybrid approach by combining multiple social metrics perform better than those using only one of these metrics [4], [14], [76]. Evidently, there is a tradeoff between the amount of bearable complexity and the performance of the network. This added complexity also brings additional delays which may not be desirable for highly mobile vehicles and intelligent transportation systems. Thus, there is a need to minimize the tradeoff between the performance and complexity of protocols.

B. STANDARDIZATION AND REDESIGNING EXISTING PROTOCOLS

The existing protocols and optimization techniques require to be redesigned for social networks. Dynamic protocols for scheduling, forwarding and queuing of messages and signals need to be designed in order to integrate social perspective with the existing protocols. For instance, in case of partial or completely unavailability of information about neighboring devices, what type of strategies must be employed for making an intelligent decision. Although some approaches using evolutionary algorithms have been investigated [86], [222], yet their efficient designing is still an unsolved issue. Standardization of these social algorithms is again a challenging issue which needs to be addressed for the successful implementation of social protocols in the network.

C. ACCURACY OF ESTIMATED SOCIAL TIES

The accuracy of how strong and weak social ties are depending on the global state information, which includes parameters such as the contact time and contact frequency. Most of the studies consider that this information is available *a priori* and it can be estimated perfectly. However, this is not the case in practical networks as the global state information usually requires the collection of a large amount of data which may not be feasible in ad-hoc networks [223]. Moreover, the data collection time is significantly longer and is not considered a feasible approach for time-varying environments. Some studies utilize self-reported techniques to acquire this information, but these techniques cannot be used to detect dynamic social ties. Hence, acquisition of accurate social relations under time-varying communication scenarios is still an open issue.

D. DETERMINATION OF APPROPRIATE METRIC

The forwarding decision in social networks is greatly dependent on the social relation of devices. Most of the social routing methods make use of either one or two social metrics to estimate the strength of social ties. However, in reality, using one or two metrics may not be enough as the topologies and environments greatly differ and are usually complex. Therefore, it is difficult to accurately estimate the social characteristics using only a few social metrics especially in the presence of highly mobile devices or in a frequently changing environment. Thus, more research efforts should be focused towards creating generic social metrics that can be applied to a wide range of environments. In addition, it is of high interest to investigate which types of social metrics are suitable for a particular application or scenario.

E. PEER DISCOVERY

One of the most neglected factors in social assistance is the discovery of peers and nearby devices. Note that the resources allocated for peer discovery are generally limited and therefore need to be utilized wisely. An illustration for resource allocation in LTE system is shown in Figure 9. Generally, peer discovery is done by sending beacons over control channels and based on the limited amount of information received by the connection-initiating device, it is difficult to estimate the feasibility of this connection from a social perspective. To partly address this concern some pre-association services can be utilized at the cost of a higher latency. However, the increased latency of pre-association schemes makes them



FIGURE 9. Device discovery resource allocation in LTE. The resources are dived into discovery resources and LTE uplink resources. Each discovery period consists of 64 subframes, wherein, one subframe forms a single discovery resource unit.

unsuitable for emergency applications. For these reasons, it is imperative to formulate a low latency and resource-efficient solution to enable social assistance in future devices.

F. REALIZING CROSS-LAYER DESIGN

One can perceive from the above arguments that social protocols are mostly MAC or Physical Layer based protocols. In addition, most of the studies consider a single layer to improve the performance of the system. The integration of these protocols with higher layers is an exciting prospect which requires a deeper investigation. Thus, more research efforts are needed to be directed towards realizing a crosslayer design for future wireless technologies such as IoT and VANETs.

G. MINIMIZATION OF FEEDBACK AND OVERHEAD

A significantly large number of network resources are consumed when connectivity needs to be maintained frequently. This connectivity comes at the cost of feedback overhead, which can consume both time and frequency resources. Replication-based services can also be used to improve the delivery ratio of the messages and, resultantly, improve the social ties. However, in a resource-constrained communication environment, replication based services use excessive power and bandwidth. This scenario gets worse when multiple intermediate devices are involved in the path from source to destination. The intermediate devices not only have to reserve resources for retransmissions but they also remain occupied until the successful transmission of messages.

H. SUMMARY AND INSIGHTS

In this section, we have highlighted the major open issues with respect to the implementation of social networking enabled 5G communications. Upon review of recent developments, we noted that the tradeoff between performance and complexity of incorporating social-awareness in wireless networks merits full clarification. Moreover, the standardization and protocol redesigning efforts are quite weak which will result in slowing down the evolution of social assisted wireless networking. Similarly, the research studies on crosslayered design and utilization of multicast protocols are limited and require the attention of the research community. Establishment of end-to-end security will be essential, especially with the deployment of ultra-dense networks. However, the security enhancement of social-aware 5G networks is still under-explored and require more research activity to develop advanced security techniques.

VI. EMERGING SOCIAL NETWORKS AND FUTURE DIRECTIONS

Based on the analysis of the above-mentioned studies, it can be inferred that social networking approaches have the potential to enhance the performance of future wireless networks. Here, we identify future prospects and possibilities for integrating the social networking approaches with emerging communication paradigms.

A. NON-ORTHOGONAL MULTIPLE ACCESS

NOMA has gained a significant amount of attention from the research community due to its promise of enhancing spectral efficiency and throughput of the users at the edge of the cell [224], [225]. All these benefits come with the prospect of minimum latency and power requirements. Moreover, the users within a cell can share the time and frequency resources by regulating their power ratio. This allows users with better wireless channel conditions to apply successive interference cancelation (SIC) technique to remove the messages for other users and subsequently decode their messages [226], [227]. The benefit of non-orthogonality in NOMA has been recently used in a number of wireless systems and applications. Also, NOMA systems generally perform better than OFDMA systems [228], [229], which signifies the usefulness of NOMA in future networks. Despite this importance of NOMA, the work exploiting social networking approaches for NOMA is almost non-existent. We anticipate that using social-aware strategies can further improve the user-fairness in NOMA. Moreover, by adopting social approaches in NOMA, the issue of trust management between resource sharing devices can be handled more effectively.

B. VISIBLE LIGHT COMMUNICATION

VLC has opened new avenues to realize ad-hoc and shortrange communication [230], [231]. It is a low-bit-rate (LBR) communication system where light-emitting diodes (LEDs) act as senders and receivers using line-of-sight medium [232]. This cost-effective solution is considered to be most suitable for establishing sensor networks and consumer electronic networks. Moreover, VLC provides the flexibility of adding nodes on the go and has the capability of supporting various multiple access protocols. It can provide data speed up to 1Gbps in a normal setting and 4Gbps in a special setting which has motivated the works in [233] and [234]. From the security perspective, the VLC can be used to materialize the concept of "Protected Zones". It is because light cannot penetrate in walls which makes VLC more secure as compared to RF communication. Due to the visibility of communication, the receivers and senders are exposed at all the time and passive eavesdropping cannot take place [235]. The limited range of VLC is also considered beneficial for minimizing the interference [236]. Also, in contrast to RF communications, there are no side effects of the VLC on human health, especially for pregnant women.

Although numerous studies exist to improve the performance of VLC networks, yet the social networking perspective of VLC is mostly missing in the literature. Due to the adhoc nature of VLC, the social approaches are directly applicable to these networks. The social-aware VLC networks can perform better by analyzing the patterns of connectivity of devices. More specifically, the LEDs can be programmed to turn on and off based on the predicted behavior of devices in order to save energy and improve network lifetime. In addition, cluster-based VLC networks can work together by using centrality and social ties to better facilitate other devices in the network. An illustration of the socially connected VLC system is given in Figure 10.

C. SMART SENSORS

Sensors have the capacity to determine key parameters and to transfer the results to communications systems that can record and ultimately analyze the outputs instigating the requisite responses in a timely manner. Such sensors are well established for analyzing the production/transfer of items (perishable/non-perishable), recording characteristics, such as location, temperature, humidity and associated storage conditions, duration of movements, integrity of packaging, exposure to certain adulterants (chemicals, gases or pathogenic organisms) and finally determining the integrity of the item and its acceptability for use [237]. The close integration of such sensing systems with communications networks can provide vital information for producers, users and officials (e.g. quality assessors and government agencies) ensuring quality and global movement monitoring. Similar approaches are envisioned for monitoring healthcare [238] where discrete or arrayed sensors can record key physiological parameters (e.g. temperature, movement, glucose levels, or disease-related biomarkers) enabling both the patient and physician measure health and well-being and instigate appropriate interventions when necessary. The integrity of the communication systems are vital for recording the data, allowing appropriate checking, and updating the associated records (e.g. health records of a patient). Such approaches greatly facilitate timely medical interventions based on early detection of disease or where a patient requires close supervision [239] due to certain clinical conditions (e.g. diabetes, cancer, cardiac diseases). Such strategies can potentially provide much better clinical health monitoring,



FIGURE 10. Demonstration of a socially connected VLC system. Three communities labelled A, B, and C each comprise of several connected devices like laptops and smartphones. Centrality is maintained by the light-sources and A, B, and C interact through line-of-sight links.

improve disease management, reduce the economic load of health provision for governments and, above all, provide a much enhanced 'quality-of-life' with people able to remain active and embedded in their own communities supported by effective technological advances.

Key items that need to be addressed relate to the maintenance of the integrity of the data, ensuring correct ethical usage and avoiding inappropriate exploitation. However, potentially much more can be achieved by integrating both sensing and communications systems. Given the huge involvement of people worldwide in such systems, the question arises as to how the activities of people 'online' can be used to determine their well-being. The nature of their activity profile, linked to sensing, may provide even more scope for disease monitoring particularly in relation to behavioral or neurological diseases. Since environmental conditions are of huge importance to some diseases (e.g. asthma), smart systems could provide much more effective help to sufferers in terms of awareness, and the need for interventions. There are many more scenarios where the integration of smart sensing systems with newly developing communications approaches can be highly beneficial but significant research is still necessary to optimize the sensing elements to supply the appropriate outputs in a cost-effective and timely manner with minimal human intervention.

D. INFORMATION CENTRIC WIRELESS NETWORKS

The aforementioned studies on ICN mainly focus on wired environments. However, the decoupling between publishers and subscribers can be utilized to increase the scale of communication. In this context, Amadeo *et al.* [240] analyzed the problem of energy efficiency in the content-centric network. They proposed to use personal mobile devices to temporarily cache the content and then distribute it in a multihop access network. Some of the potential benefits of content-centric MANETs are discussed in [241]. In a similar work, Shim and Kim [242] proposed a data aggregation strategy in ICN based wireless sensor networks. The same authors proposed to use IP-based communication architecture instead of ID-based ICN to reduce energy consumption.

A tree-based data aggregation scheme for ICN sensor networks was proposed in [243] and [244]. Jin *et al.* [245] provided a model to reduce energy consumption by exploiting the publisher/ subscriber paradigm. The proposed model not only reduces the energy consumed by sensor nodes but also satisfies the decoupling principles. Some studies have also proposed virtualization techniques [246], [247] in wireless ICNs. However, still, the work on social aspects of wireless ICNs is almost non-existent. Based on the discussion of potential applications of social networks in the above sections, one can predict that content dissemination can be further improved by using social characteristics of wireless devices in ICN. Thus, there is a need to design social models and incentive-based techniques to use the wireless contentbased networks at their fullest potential.

E. SMART GRIDS

Smart grids are considered to revolutionize the energy infrastructure of the future smart cities. This technology aims to provide efficient, sustainable and cost-effective power generation solutions [248]. However, it is a challenging task to handle a large amount of data generated by smart meters to provide precise predictions. Commonly used data mining algorithms may not be applicable to provide accurate predictions for such a large amount of data. Social network relationships can help to effectively address this problem by digging a large number of hidden information [249].

Although a limited work [249]–[253] exists on exploring social aspects of the smart grid, yet these studies either review the advancement in smart grids or focus on predicting the monthly expenditure of consumers. Some learning techniques have also been developed better analyze the social relationship among smart meters [254], [255], however, the studies on security, grid management and demand response of social smart grid networks are absent. Also, the work on route management using social-aware communication between grid station and smart meters is missing. We expect that by solving these issues using social networking approaches, substantial gains can be obtained in terms of energy efficiency. Social aspects of smart grid communication can also help to improve the fault tolerance mechanism of the smart grid. Social ties can also be used by smart meters to monitor intermittent connectivity among meters. In case of any fluctuations or instability in connectivity, the neighboring meters can inform the main grid station to take further actions.

F. MMWAVE COMMUNICATION

Due to rapid growth in demand for data traffic, researchers have started exploring alternative procedures and methods to extend the bandwidth of wireless systems. In this regard, exploitation of millimeter wave (mmWave) frequencies (10-300 GHz) is considered to be a viable option to overcome the bandwidth limitations of existing networks. Due to small wavelengths of this band, large arrays of antennas can be used to provide higher directivity and improved signal strength at the receiver [256]. Higher directivity can be achieved by using beamforming approaches which also act effectively against path losses [257]. Also, this high capacity will allow connectivity of a large number of devices. Therefore, there is a growing consensus that mmWave communication will play a key role in the forthcoming generation of networks.

Although previously mentioned studies on mmWaves have laid a solid foundation, yet the application of social networking approaches in mmWave is still in its fancy. Moreover, exactly how much performance gains can be obtained, using social networking with mmWave communication, is still unknown. Lately, social schemes have been used to provide cell association services and dynamic clustering algorithms in dense networks [258]–[260]. However, the applicability of these techniques has not yet been exploited for a purely mmWave enabled cellular environment. In addition to this, social-aware protocols can be used to provide contextual information of devices which can minimize co-tier and cross-tier interference in mmWave cellular infrastructure.

G. SUMMARY AND INSIGHTS

This section has provided a discussion on some of the main communication domains that can benefit from adopting social networking approaches. These technologies include NOMA, VLC, smart sensors, wireless ICN, smart grid, and mmWave communication. Incremental incorporation of social networking approach in these technologies can constructively alter the perspective of academics and researchers on the design of the next generation of communication networks.

VII. CONCLUSION

We presented current state-of-the-art and the future developments of social networking approaches for 5G communications. Specifically, the paper offered a thorough discussion on the architecture, characteristics, and vulnerabilities of social networks. We have also focused on various applications of social networks for 5G technologies. This was followed by a discussion on recent developments in technologies such as D2D, IoT, cloud computing and vehicular networks. This study has identified that more work needs to be done to address the vulnerabilities in existing social networks. Also, there is a need to evolve universally acceptable performance metrics for social networking applied to communications. Finally, the current hardware limitations and lack of concentrated standardization efforts must be addressed for a successful incorporation of social networking in future communications technologies.

REFERENCES

- M. Amadeo *et al.*, "Information-centric networking for the Internet of Things: Challenges and opportunities," *IEEE Netw.*, vol. 30, no. 2, pp. 92–100, Mar./Apr. 2016.
- [2] Z. D. Stephens *et al.*, "Big data: Astronomical or genomical?" *PLoS Biol.*, vol. 13, no. 7, p. e1002195, 2015.
- [3] R. M. Bond *et al.*, "A 61-million-person experiment in social influence and political mobilization," *Nature*, vol. 489, no. 7415, pp. 295–298, 2012.
- [4] P. Hui, J. Crowcroft, and E. Yoneki, "BUBBLE rap: Social-based forwarding in delay-tolerant networks," *IEEE Trans. Mobile Comput.*, vol. 10, no. 11, pp. 1576–1589, Nov. 2011.
- [5] B. Han, P. Hui, V. S. A. Kumar, M. V. Marathe, J. Shao, and A. Srinivasan, "Mobile data offloading through opportunistic communications and social participation," *IEEE Trans. Mobile Comput.*, vol. 11, no. 5, pp. 821–834, May 2012.
- [6] N. Kayastha, D. Niyato, P. Wang, and E. Hossain, "Applications, architectures, and protocol design issues for mobile social networks: A survey," *Proc. IEEE*, vol. 99, no. 12, pp. 2130–2158, Dec. 2011.
- [7] Y. Zhu, B. Xu, X. Shi, and Y. Wang, "A survey of social-based routing in delay tolerant networks: Positive and negative social effects," *IEEE Commun. Surveys Tuts.*, vol. 15, no. 1, pp. 387–401, 1st Quart., 2013.
- [8] K. Wei, X. Liang, and K. Xu, "A survey of social-aware routing protocols in delay tolerant networks: Applications, taxonomy and design-related issues," *IEEE Commun. Surveys Tuts.*, vol. 16, no. 1, pp. 556–578, 1st Quart., 2014.
- [9] S. Singh, N. Saxena, A. Roy, and H. Kim, "A survey on 5G network technologies from social perspective," *IETE Tech. Rev.*, vol. 34, no. 1, pp. 30–39, 2017.
- [10] Y. Li, T. Wu, P. Hui, D. Jin, and S. Chen, "Social-aware D2D communications: Qualitative insights and quantitative analysis," *IEEE Commun. Mag.*, vol. 52, no. 6, pp. 150–158, Jun. 2014.
- [11] I. Lequerica, M. G. Longaron, and P. M. Ruiz, "Drive and share: Efficient provisioning of social networks in vehicular scenarios," *IEEE Commun. Mag.*, vol. 48, no. 11, pp. 90–97, Nov. 2010.

- [12] V. Krishnamurthy and H. V. Poor, "A tutorial on interactive sensing in social networks," *IEEE Trans. Comput. Social Syst.*, vol. 1, no. 1, pp. 3–21, Mar. 2014.
- [13] E. Miluzzo et al., "Sensing meets mobile social networks: The design, implementation and evaluation of the cenceme application," in Proc. 6th ACM Conf. Embedded Netw. Sensor Syst., 2008, pp. 337–350.
- [14] E. M. Daly and M. Haahr, "Social network analysis for routing in disconnected delay-tolerant manets," in *Proc. 8th ACM Int. Symp. Mobile Ad Hoc Netw. Comput.*, 2007, pp. 32–40.
- [15] A. M. Ortiz, D. Hussein, S. Park, S. N. Han, and N. Crespi, "The cluster between Internet of Things and social networks: Review and research challenges," *IEEE Internet Things J.*, vol. 1, no. 3, pp. 206–215, Jun. 2014.
- [16] A. K. Triantafyllidis, C. Velardo, D. Salvi, S. A. Shah, V. G. Koutkias, and L. Tarassenko, "A survey of mobile phone sensing, self-reporting, and social sharing for pervasive healthcare," *IEEE J. Biomed. Health Inform.*, vol. 21, no. 1, pp. 218–227, Jan. 2017.
- [17] X. Zhang et al., "Incentives for mobile crowd sensing: A survey," IEEE Commun. Surveys Tuts., vol. 18, no. 1, pp. 54–67, 1st Quart., 2016.
- [18] J. Luo, J. Zhang, Y. Cui, L. Yu, and X. Wang, "Asymptotic analysis on content placement and retrieval in manets," *IEEE/ACM Trans. Netw.*, vol. 25, no. 2, pp. 1103–1118, Jan. 2017.
- [19] Y.-H. Kao, B. Krishnamachari, M.-R. Ra, and F. Bai, "Hermes: Latency optimal task assignment for resource-constrained mobile computing," *IEEE Trans. Mobile Comput.*, vol. 16, no. 11, pp. 3056–3069, 2017.
- [20] K. Gai, M. Qiu, L. Tao, and Y. Zhu, "Intrusion detection techniques for mobile cloud computing in heterogeneous 5G," *Secur. Commun. Netw.*, vol. 9, no. 16, pp. 3049–3058, 2016.
- [21] A. Anderson, D. Huttenlocher, J. Kleinberg, and J. Leskovec, "Effects of user similarity in social media," in *Proc. 5th ACM Int. Conf. Web Search Data Mining*, 2012, pp. 703–712.
- [22] C. R. Shalizi and A. C. Thomas, "Homophily and contagion are generically confounded in observational social network studies," *Sociol. Meth*ods Res., vol. 40, no. 2, pp. 211–239, 2011.
- [23] J. Li, R. Li, and J. Kato, "Future trust management framework for mobile ad hoc networks," *IEEE Commun. Mag.*, vol. 46, no. 4, pp. 108–114, Apr. 2008.
- [24] S. Aral and D. Walker, "Identifying influential and susceptible members of social networks," *Science*, vol. 337, no. 6092, pp. 337–341, 2012.
- [25] D. J. Watts and S. H. Strogatz, "Collective dynamics of 'small-world' networks," *Nature*, vol. 393, no. 6684, pp. 440–442, 1998.
- [26] Q. Xu, Z. Su, K. Zhang, P. Ren, and X. S. Shen, "Epidemic information dissemination in mobile social networks with opportunistic links," *IEEE Trans. Emerg. Topics Comput.*, vol. 3, no. 3, pp. 399–409, Sep. 2015.
- [27] Z. Li, C. Wang, S. Yang, C. Jiang, and X. Li, "LASS: Local-activity and social-similarity based data forwarding in mobile social networks," *IEEE Trans. Parallel Distrib. Syst.*, vol. 26, no. 1, pp. 174–184, Jan. 2015.
- [28] C. Dellarocas, "Mechanisms for coping with unfair ratings and discriminatory behavior in online reputation reporting systems," in *Proc. 21st Int. Conf. Inf. Syst.*, 2000, pp. 520–525.
- [29] Y. Wang *et al.*, "CATrust: Context-aware trust management for serviceoriented ad hoc networks," *IEEE Trans. Services Comput.*, to be published, doi: 10.1109/TSC.2016.2587259.
- [30] D. Wu, F. Zhang, H. Wang, and R. Wang, "Security-oriented opportunistic data forwarding in mobile social networks," *Future Gener. Comput. Syst.*, vol. 87, pp. 803–815, Oct. 2017.
- [31] D. Qin, S. Yang, S. Jia, Y. Zhang, J. Ma, and Q. Ding, "Research on trust sensing based secure routing mechanism for wireless sensor network," *IEEE Access*, vol. 5, pp. 9599–9609, 2017.
- [32] R. Chen, F. Bao, and J. Guo, "Trust-based service management for social Internet of Things systems," *IEEE Trans. Dependable Secure Comput.*, vol. 13, no. 6, pp. 684–696, Nov. 2016.
- [33] H. Rowaihy, W. Enck, P. McDaniel, and T. La Porta, "Limiting Sybil attacks in structured P2P networks," in *Proc. Int. Conf. Comput. Commun. (INFOCOM)*, May 2007, pp. 2596–2600.
- [34] S. Abbas, M. Merabti, D. Llewellyn-Jones, and K. Kifayat, "Lightweight sybil attack detection in MANETs," *IEEE Syst. J.*, vol. 7, no. 2, pp. 236–248, Jun. 2013.
- [35] F. Gai, J. Zhang, P. Zhu, and X. Jiang, "Trust on the ratee: A trust management system for social Internet of vehicles," *Wireless Commun. Mobile Comput.*, vol. 2017, 2017, Art. no. 7089259.

- [36] Y. Wang, I.-R. Chen, J.-H. Cho, and J. J. Tsai, "Trust-based task assignment with multiobjective optimization in service-oriented ad hoc networks," *IEEE Trans. Netw. Service Manage.*, vol. 14, no. 1, pp. 217–232, Mar. 2017.
- [37] C. J. Fung, J. Zhang, I. Aib, and R. Boutaba, "Robust and scalable trust management for collaborative intrusion detection," in *Proc. Int. Symp. Integr. Netw. Manage.*, 2009, pp. 33–40.
- [38] Y. Wu et al., "Challenges of mobile social device caching," *IEEE Access*, vol. 4, pp. 8938–8947, 2016.
- [39] M. Yamamoto, "A survey of caching networks in content oriented networks," *IEICE Trans. Commun.*, vol. 99, no. 5, pp. 961–973, 2016.
- [40] E. Bastug, J.-L. Guénégo, and M. Debbah, "Proactive small cell networks," in *Proc. Int. Conf. Telecommun. (ICT)*, 2013, pp. 1–5.
- [41] E. Bastug, M. Bennis, and M. Debbah, "Living on the edge: The role of proactive caching in 5G wireless networks," *IEEE Commun. Mag.*, vol. 52, no. 8, pp. 82–89, Aug. 2014.
- [42] E. Bastug, M. Bennis, and M. Debbah, "Anticipatory caching in small cell networks: A transfer learning approach," in *Proc. 1st KuVS Workshop Anticipatory Netw.*, 2014, pp. 1–3.
- [43] V. A. Siris, X. Vasilakos, and D. Dimopoulos, "Exploiting mobility prediction for mobility & popularity caching and dash adaptation," in *Proc. Int. Symp. World Wireless, Mobile Multimedia Netw. (WoWMoM)*, 2016, pp. 1–8.
- [44] J. Tadrous and A. Eryilmaz, "On optimal proactive caching for mobile networks with demand uncertainties," *IEEE/ACM Trans. Netw.*, vol. 24, no. 5, pp. 2715–2727, Oct. 2016.
- [45] J. Tadrous, A. Eryilmaz, and H. El Gamal, "Joint smart pricing and proactive content caching for mobile services," *IEEE/ACM Trans. Netw.*, vol. 24, no. 4, pp. 2357–2371, Aug. 2016.
- [46] C. Yi, S. Huang, and J. Cai, "An incentive mechanism integrating joint power, channel and link management for social-aware D2D content sharing and proactive caching," *IEEE Trans. Mobile Comput.*, vol. 17, no. 4, pp. 789–802, Apr. 2018.
- [47] D. Wu, D. I. Arkhipov, T. Przepiorka, Q. Liu, J. A. McCann, and A. C. Regan, "DeepOpp: Context-aware mobile access to social media content on underground metro systems," in *Proc. IEEE 37th Int. Conf. Distrib. Comput. Syst. (ICDCS)*, Jun. 2017, pp. 1219–1229.
- [48] K. Machado, A. Boukerche, E. Cerqueira, and A. A. Loureiro, "A socially-aware in-network caching framework for the next generation of wireless networks," *IEEE Commun. Mag.*, vol. 55, no. 12, pp. 38–43, Dec. 2017.
- [49] O. Semiari, W. Saad, S. Valentin, M. Bennis, and H. V. Poor, "Contextaware small cell networks: How social metrics improve wireless resource allocation," *IEEE Trans. Wireless Commun.*, vol. 14, no. 11, pp. 5927–5940, Nov. 2015.
- [50] O. Semiari, W. Saad, and M. Bennis, "Context-aware scheduling of joint millimeter wave and microwave resources for dual-mode base stations," in *Proc. Int. Conf. Commun. (ICC)*, May 2016, pp. 1–6.
- [51] N. Namvar, W. Saad, B. Maham, and S. Valentin, "A context-aware matching game for user association in wireless small cell networks," in *Proc. Int. Conf. Acoust., Speech Signal Process. (ICASSP)*, 2014, pp. 439–443.
- [52] Y. Zhang, E. Pan, L. Song, W. Saad, Z. Dawy, and Z. Han, "Social network aware device-to-device communication in wireless networks," *IEEE Trans. Wireless Commun.*, vol. 14, no. 1, pp. 177–190, Jan. 2015.
- [53] V. Shah and A. O. Fapojuwo, "Socially aware resource allocation for device-to-device communication in downlink OFDMA networks," in *Proc. Wireless Commun. Netw. Conf. (WCNC)*, Apr. 2014, pp. 1673–1678.
- [54] Y. Li, S. Su, and S. Chen, "Social-aware resource allocation for device-todevice communications underlaying cellular networks," *IEEE Wireless Commun. Lett.*, vol. 4, no. 3, pp. 293–296, Jun. 2015.
- [55] P. Zhao, L. Feng, P. Yu, W. Li, and X. Qiu, "A social-aware resource allocation for 5g device-to-device multicast communication," *IEEE Access*, vol. 5, pp. 15717–15730, 2017.
- [56] J. Zhao, Y. Liu, K. K. Chai, M. Elkashlan, and Y. Chen, "Matching with peer effects for context-aware resource allocation in D2D communications," *IEEE Commun. Lett.*, vol. 21, no. 4, pp. 837–840, Apr. 2017.
- [57] P. Floréen, P. Kaski, V. Polishchuk, and J. Suomela, "Almost stable matchings by truncating the gale-shapley algorithm," *Algorithmica*, vol. 58, no. 1, pp. 102–118, 2010.

- [58] Y. Zhang, L. Song, C. Jiang, N. H. Tran, Z. Dawy, and Z. Han, "A socialaware framework for efficient information dissemination in wireless ad hoc networks," *IEEE Commun. Mag.*, vol. 55, no. 1, pp. 174–179, Jan. 2017.
- [59] Z. Su, Q. Xu, M. Fei, and M. Dong, "Game theoretic resource allocation in media cloud with mobile social users," *IEEE Trans. Multimedia*, vol. 18, no. 8, pp. 1650–1660, Aug. 2016.
- [60] N. H. Tran, D. H. Tran, S. Ren, Z. Han, E.-N. Huh, and C. S. Hong, "How geo-distributed data centers do demand response: A game-theoretic approach," *IEEE Trans. Smart Grid*, vol. 7, no. 2, pp. 937–947, 2016.
- [61] F. Xia, L. Liu, J. Li, J. Ma, and A. V. Vasilakos, "Socially aware networking: A survey," *IEEE Syst. J.*, vol. 9, no. 3, pp. 904–921, Sep. 2015.
- [62] Y. Li, P. Luo, Z.-P. Fan, K. Chen, and J. Liu, "A utility-based link prediction method in social networks," *Eur. J. Oper. Res.*, vol. 260, no. 2, pp. 693–705, 2017.
- [63] C. A. Bliss, M. R. Frank, C. M. Danforth, and P. S. Dodds, "An evolutionary algorithm approach to link prediction in dynamic social networks," *J. Comput. Sci.*, vol. 5, no. 5, pp. 750–764, 2014.
- [64] Y.-L. He, J. N. Liu, Y.-X. Hu, and X.-Z. Wang, "OWA operator based link prediction ensemble for social network," *Expert Syst. Appl.*, vol. 42, no. 1, pp. 21–50, 2015.
- [65] J. Valverde-Rebaza and A. de Andrade Lopes, "Exploiting behaviors of communities of twitter users for link prediction," *Social Netw. Anal. Mining*, vol. 3, no. 4, pp. 1063–1074, 2013.
- [66] R.-H. Li, J. X. Yu, and J. Liu, "Link prediction: The power of maximal entropy random walk," in *Proc. 20th ACM Int. Conf. Inf. Knowl. Manage.*, 2011, pp. 1147–1156.
- [67] H. Liu, Z. Hu, H. Haddadi, and H. Tian, "Hidden link prediction based on node centrality and weak ties," *EPL (Europhys. Lett.)*, vol. 101, no. 1, p. 18004, 2013.
- [68] B. Qiu, K. Ivanova, J. Yen, and P. Liu, "Behavior evolution and eventdriven growth dynamics in social networks," in *Proc. 2nd Int. Conf. Social Comput. (SocialCom)*, 2010, pp. 217–224.
- [69] S.-H. Yang, B. Long, A. Smola, N. Sadagopan, Z. Zheng, and H. Zha, "Like like alike: Joint friendship and interest propagation in social networks," in *Proc. 20th Int. Conf. World Wide Web*, 2011, pp. 537–546.
- [70] B. Moradabadi and M. R. Meybodi, "Link prediction based on temporal similarity metrics using continuous action set learning automata," *Phys. A, Stat. Mech. Appl.*, vol. 460, pp. 361–373, Oct. 2016.
- [71] İ Günes, Ş. Gündüz-Ögüdücü, and Z. Çataltepe, "Link prediction using time series of neighborhood-based node similarity scores," *Data Mining Knowl. Discovery*, vol. 30, no. 1, pp. 147–180, 2016.
- [72] A. Lebedev, J. Lee, V. Rivera, and M. Mazzara, "Link prediction using top-k shortest distances," in *Proc. Brit. Int. Conf. Databases*. London, U.K.: Springer, 2017, pp. 101–105.
- [73] Y. Cao and Z. Sun, "Routing in delay/disruption tolerant networks: A taxonomy, survey and challenges," *IEEE Commun. Surveys Tuts.*, vol. 15, no. 2, pp. 654–677, 2nd Quart., 2013.
- [74] S. Yang, X. Yang, C. Zhang, and E. Spyrou, "Using social network theory for modeling human mobility," *IEEE Netw.*, vol. 24, no. 5, pp. 6–13, Sep./Oct. 2010.
- [75] P. Hui and J. Crowcroft, "How small labels create big improvements," in *Proc. Int. Conf. Pervasive Comput. Commun. Workshops*, Mar. 2007, pp. 65–70.
- [76] E. Bulut and B. K. Szymanski, "Friendship based routing in delay tolerant mobile social networks," in *Proc. Global Telecommun. Conf. (GLOBE-COM)*, Dec. 2010, pp. 1–5.
- [77] M. Musolesi, P. Hui, C. Mascolo, and J. Crowcroft, "Writing on the clean slate: Implementing a socially-aware protocol in haggle," in *Proc. Int. Symp. World Wireless, Mobile Multimedia Netw.*, Jun. 2008, pp. 1–6.
- [78] E. M. Daly and M. Haahr, "Social network analysis for information flow in disconnected delay-tolerant MANETs," *IEEE Trans. Mobile Comput.*, vol. 8, no. 5, pp. 606–621, May 2009.
- [79] C. Boldrini, M. Conti, and A. Passarella, "Exploiting users social relations to forward data in opportunistic networks: The HiBoP solution," *Pervasive Mobile Comput.*, vol. 4, no. 5, pp. 633–657, 2008.
- [80] Q. Yuan, I. Cardei, and J. Wu, "Predict and relay: An efficient routing in disruption-tolerant networks," in *Proc. 10th ACM Int. Symp. Mobile Ad Hoc Netw. Comput.*, 2009, pp. 95–104.
- [81] S. Patra, S. Saha, V. Shah, S. Sengupta, K. G. Singh, and S. Nandi, "A qualitative survey on multicast routing in delay tolerant networks," in *Recent Trends in Wireless and Mobile Networks*. Springer, 2011, pp. 197–206.

- [82] M. Radenkovic and A. Grundy, "Efficient and adaptive congestion control for heterogeneous delay-tolerant networks," *Ad Hoc Netw.*, vol. 10, no. 7, pp. 1322–1345, 2012.
- [83] W. Gao, Q. Li, B. Zhao, and G. Cao, "Social-aware multicast in disruption-tolerant networks," *IEEE/ACM Trans. Netw.*, vol. 20, no. 5, pp. 1553–1566, Oct. 2012.
- [84] M. Radenkovic and A. Grundy, "Congestion aware forwarding in delay tolerant and social opportunistic networks," in *Proc. Int. Conf. Wireless Demand Netw. Syst. Services (WONS)*, Jan. 2011, pp. 60–67.
- [85] P. Yuan, H. Ma, and P. Duan, "On exploiting few strangers for data forwarding in delay tolerant networks," in *Proc. Int. Conf. Mobile Ad-Hoc Sensor Netw. (MSN)*, 2011, pp. 282–288.
- [86] E. R. Silva and P. R. Guardieiro, "An efficient genetic algorithm for anycast routing in delay/disruption tolerant networks," *IEEE Commun. Lett.*, vol. 14, no. 4, pp. 315–317, Apr. 2010.
- [87] F. Jiang, Y. Liu, C. Li, and C. Sun, "Energy-efficient multicast transmission for underlay device-to-device communications: A socialaware perspective," *Mobile Inf. Syst.*, vol. 2017, 2017, Art. no. 7148564.
- [88] A. Orsino, L. Militano, G. Araniti, and A. Iera, "Social-aware content delivery with D2D communications support for emergency scenarios in 5G systems," in *Proc. 22th Eur. Conf. Eur. Wireless*, 2016, pp. 1–6.
- [89] G. Zhang, K. Yang, and H. H. Chen, "Socially aware cluster formation and radio resource allocation in D2D networks," *IEEE Wireless Commun.*, vol. 23, no. 4, pp. 68–73, Aug. 2016.
- [90] M. Ali, T. Suleman, and Z. A. Uzmi, "MMAC: A mobility-adaptive, collision-free MAC protocol for wireless sensor networks," in *Proc. IPCCC*, vol. 28, 2005, no. 2, pp. 401–407.
- [91] W. Wanalertlak, B. Lee, C. Yu, M. Kim, S.-M. Park, and W.-T. Kim, "Behavior-based mobility prediction for seamless handoffs in mobile wireless networks," *Wireless Netw.*, vol. 17, no. 3, pp. 645–658, 2011.
- [92] F. Calabrese, G. Di Lorenzo, and C. Ratti, "Human mobility prediction based on individual and collective geographical preferences," in *Proc. Int. Conf. Intell. Transp. Syst. (ITSC)*, 2010, pp. 312–317.
- [93] H. Bapierre, G. Groh, and S. Theiner, "A variable order Markov model approach for mobility prediction," *Proc. STAMI Int. Joint Conf. Artif. Intell.*, Barcelona, Spain, Jul. 2011, pp. 8–16.
- [94] Z. Yi, X. Dong, X. Zhang, and W. Wang, "Spatial traffic prediction for wireless cellular system based on base stations social network," in *Proc. Annu. IEEE Syst. Conf. (SysCon)*, Apr. 2016, pp. 1–5.
- [95] G. I. Tsiropoulos, D. G. Stratogiannis, N. Mantas, and M. Louta, "The impact of social distance on utility based resource allocation in next generation networks," in *Proc. Int. Congr. Ultra Mod. Telecommun. Control Syst. Workshops (ICUMT)*, 2011, pp. 1–6.
- [96] Z. Ning *et al.*, "A cooperative quality-aware service access system for social Internet of vehicles," *IEEE Internet Things J.*, vol. 5, no. 4, pp. 2506–2517, Aug. 2017.
- [97] Q. Yuan, W. Zhang, C. Zhang, X. Geng, G. Cong, and J. Han, "PRED: Periodic region detection for mobility modeling of social media users," in *Proc. 10th ACM Int. Conf. Web Search Data Mining*, 2017, pp. 263–272.
- [98] J.-K. Lee and J. C. Hou, "Modeling steady-state and transient behaviors of user mobility: Formulation, analysis, and application," in *Proc. 7th ACM Int. Symp. Mobile Netw. Comput.*, 2006, pp. 85–96.
- [99] S. Gambs, M.-O. Killijian, and M. N. del Prado Cortez, "Next place prediction using mobility Markov chains," in *Proc. 1st Workshop Meas.*, *Privacy, Mobility*, 2012, p. 3.
- [100] M. Al-Hattab, M. Takruri, and J. Agbinya, "Mobility prediction using pattern matching," *Int. J. Elect. Comput. Sci.*, vol. 12, no. 3, pp. 18–24, 2012.
- [101] A. Sadilek, H. Kautz, and J. P. Bigham, "Finding your friends and following them to where you are," in *Proc. 5th ACM Int. Conf. Web Search Data Mining*, 2012, pp. 723–732.
- [102] W. Mathew, R. Raposo, and B. Martins, "Predicting future locations with hidden markov models," in *Proc. ACM Conf. Ubiquitous Comput.*, 2012, pp. 911–918.
- [103] A. Monreale, F. Pinelli, R. Trasarti, and F. Giannotti, "WhereNext: A location predictor on trajectory pattern mining," in *Proc. 15th ACM SIGKDD Int. Conf. Knowl. Discovery Data Mining*, 2009, pp. 637–646.
- [104] R.-I. Ciobanu, R.-C. Marin, C. Dobre, V. Cristea, and C. X. Mavromoustakis, "ONSIDE: Socially-aware and interest-based dissemination in opportunistic networks," in *Proc. Netw. Oper. Manage. Symp. (NOMS)*, 2014, pp. 1–6.
- [105] R. I. Ciobanu, C. Dobre, and V. Cristea, "SPRINT: Social predictionbased opportunistic routing," in *Proc. Symp. Workshops World Wireless*, *Mobile Multimedia Netw. (WoWMoM)*, Jun. 2013, pp. 1–7.

- [106] H. Zhou, L. Tong, S. Xu, C. Huang, and J. Fan, "Predicting temporal centrality in opportunistic mobile social networks based on social behavior of people," *Pervasive Ubiquitous Comput.*, vol. 20, no. 6, pp. 885–897, 2016.
- [107] C. Yin, L. Gao, and S. Cui, "Scaling laws for overlaid wireless networks: A cognitive radio network versus a primary network," *IEEE/ACM Trans. Netw.*, vol. 18, no. 4, pp. 1317–1329, Aug. 2010.
- [108] L. Gao, R. Zhang, C. Yin, and S. Cui, "Throughput and delay scaling in supportive two-tier networks," *IEEE J. Sel. Areas Commun.*, vol. 30, no. 2, pp. 415–424, Feb. 2012.
- [109] J. Zhang, L. Fu, and X. Wang, "Asymptotic analysis on secrecy capacity in large-scale wireless networks," *IEEE/ACM Trans. Netw.*, vol. 22, no. 1, pp. 66–79, Feb. 2014.
- [110] Y. Chen et al., "Secrecy capacity scaling of large-scale cognitive networks," in Proc. 15th ACM Int. Symp. Mobile Ad Hoc Netw. Comput., 2014, pp. 125–134.
- [111] F. Jameel, F. Khan, M. A. A. Haider, and A. U. Haq, "Secrecy analysis of relay assisted device-to-device systems under channel uncertainty," in *Proc. Int. Conf. Frontiers Inf. Technol. (FIT)*, Dec. 2017, pp. 345–349.
- [112] F. Jameel, S. Wyne, and Z. Ding, "Secure communications in three-step two-way energy harvesting DF relaying," *IEEE Commun. Lett.*, vol. 22, no. 1, pp. 308–311, Feb. 2018.
- [113] L. Tang, H. Chen, and Q. Li, "Social tie based cooperative jamming for physical layer security," *IEEE Commun. Lett.*, vol. 19, no. 10, pp. 1790–1793, Oct. 2015.
- [114] J. Y. Ryu, J. Lee, and T. Q. S. Quek, "Confidential cooperative communication with trust degree of potential eavesdroppers," *IEEE Trans. Wireless Commun.*, vol. 15, no. 6, pp. 3823–3836, Jun. 2016.
- [115] L. Huang, X. Fan, Y. Huo, C. Hu, Y. Tian, and J. Qian, "A novel cooperative jamming scheme for wireless social networks without known CSI," *IEEE Access*, vol. 5, pp. 26476–26486, 2017.
- [116] N. Vastardis and K. Yang, "Mobile social networks: Architectures, social properties, and key research challenges," *IEEE Commun. Surveys Tuts.*, vol. 15, no. 3, pp. 1355–1371, 3rd Quart., 2013.
- [117] K. Zheng *et al.*, "Secrecy capacity scaling of large-scale networks with social relationships," *IEEE Trans. Veh. Technol.*, vol. 66, no. 3, pp. 2688–2702, Mar. 2017.
- [118] X. Chen, B. Proulx, X. Gong, and J. Zhang, "Social trust and social reciprocity based cooperative D2D communications," in *Proc. 14th ACM Int. Symp. Mobile Ad Hoc Netw. Comput.*, 2013, pp. 187–196.
- [119] L. Wang and H. Wu, "Jamming partner selection for maximising the worst D2D secrecy rate based on social trust," *Trans. Emerg. Telecommun. Technol.*, vol. 28, no. 2, p. e2992, 2017.
- [120] L. Wang, H. Wu, and G. L. Stüber, "Cooperative jamming-aided secrecy enhancement in P2P communications with social interaction constraints," *IEEE Trans. Veh. Technol.*, vol. 66, no. 2, pp. 1144–1158, Feb. 2017.
- [121] J. G. Andrews et al., "What will 5G be?" IEEE J. Sel. Areas Commun., vol. 32, no. 6, pp. 1065–1082, Jun. 2014.
- [122] "Understanding 5G: Perspectives on future technological advancements in mobile," GSMA Intell., White Paper, 2014, pp. 1–26.
- [123] M. Agiwal, A. Roy, and N. Saxena, "Next generation 5G wireless networks: A comprehensive survey," *IEEE Commun. Surveys Tuts.*, vol. 18, no. 3, pp. 1617–1655, 3rd Quart., 2016.
- [124] X. Chen, B. Proulx, X. Gong, and J. Zhang, "Exploiting social ties for cooperative D2D communications: A mobile social networking case," *IEEE/ACM Trans. Netw.*, vol. 23, no. 5, pp. 1471–1484, Oct. 2015.
- [125] Y. Zhao et al., "Social-aware resource allocation for device-to-device communications underlaying cellular networks," *IEEE Trans. Wireless Commun.*, vol. 14, no. 12, pp. 6621–6634, Dec. 2015.
- [126] F. Wang, Y. Li, Z. Wang, and Z. Yang, "Social-community-aware resource allocation for D2D communications underlaying cellular networks," *IEEE Trans. Veh. Technol.*, vol. 65, no. 5, pp. 3628–3640, May 2016.
- [127] S.-L. Chao, H.-Y. Lee, C.-C. Chou, and H.-Y. Wei, "Bio-inspired proximity discovery and synchronization for D2D communications," *IEEE Commun. Lett.*, vol. 17, no. 12, pp. 2300–2303, Dec. 2013.
- [128] H. Tang, Z. Ding, and B. C. Levy, "Enabling D2D communications through neighbor discovery in LTE cellular networks," *IEEE Trans. Signal Process.*, vol. 62, no. 19, pp. 5157–5170, Oct. 2014.
- [129] B. Zhang, Y. Li, D. Jin, and Z. Han, "Network science approach for device discovery in mobile device-to-device communications," *IEEE Trans. Veh. Technol.*, vol. 65, no. 7, pp. 5665–5679, Jul. 2016.

- [130] M. Hasan and E. Hossain, "Distributed resource allocation for relay-aided device-to-device communication under channel uncertainties: A stable matching approach," *IEEE Trans. Commun.*, vol. 63, no. 10, pp. 3882–3897, Oct. 2015.
- [131] A. Ometov et al., "Toward trusted, social-aware D2D connectivity: Bridging across the technology and sociality realms," *IEEE Wireless Commun.*, vol. 23, no. 4, pp. 103–111, Aug. 2016.
- [132] Y. Cao, T. Jiang, X. Chen, and J. Zhang, "Social-aware video multicast based on device-to-device communications," *IEEE Trans. Mobile Comput.*, vol. 15, no. 6, pp. 1528–1539, Jun. 2016.
- [133] P. K. Mishra and S. Pandey, "A method for network assisted relay selection in device to device communication for the 5G," *Int. J. Appl. Eng. Res.*, vol. 11, no. 10, pp. 7125–7131, 2016.
- [134] J. Sun, R. Zhang, and Y. Zhang, "Privacy-preserving spatiotemporal matching for secure device-to-device communications," *IEEE Internet Things J.*, vol. 3, no. 6, pp. 1048–1060, Dec. 2016.
- [135] M. Liu, J. Li, T. Liu, and Y. Chen, "Social-aware data caching mechanism in D2D-enabled cellular networks," in *Proc. Int. Conf. 5G Future Wireless Netw.* Beijing, China: Springer, 2017, pp. 650–662.
- [136] J. Li et al., "On social-aware content caching for D2D-enabled cellular networks with matching theory," *IEEE Internet Things J.*, to be published.
- [137] R. Lu, "Security and privacy preservation in vehicular social networks," UWSpace, 2012.
- [138] W. Sha, D. Kwak, B. Nath, and L. Iftode, "Social vehicle navigation: Integrating shared driving experience into vehicle navigation," in *Proc.* 14th Workshop Mobile Comput. Syst. Appl., 2013, p. 16.
- [139] S. Smaldone, L. Han, P. Shankar, and L. Iftode, "RoadSpeak: Enabling voice chat on roadways using vehicular social networks," in *Proc. 1st Workshop Social Netw. Syst.*, 2008, pp. 43–48.
- [140] M. Nitti, R. Girau, A. Floris, and L. Atzori, "On adding the social dimension to the Internet of vehicles: Friendship and middleware," in *Proc. Int. Black Sea Conf. Commun. Netw. (BlackSeaCom)*, 2014, pp. 134–138.
- [141] V. Smailovic and V. Podobnik, "Bfriend: Context-aware ad-hoc social networking for mobile users," in *Proc. 35th Int. Conv. (MIPRO)*, May 2012, pp. 612–617.
- [142] S. Gillani, F. Shahzad, A. Qayyum, and R. Mehmood, "A survey on security in vehicular ad hoc networks," in *Proc. Int. Workshop Commun. Technol. Vehicles.* Offenburg, Germany: Springer, 2013, pp. 59–74.
- [143] R. Lu, X. Lin, and X. Shen, "SPRING: A social-based privacy-preserving packet forwarding protocol for vehicular delay tolerant networks," in *Proc. IEEE INFOCOM*, 2010, pp. 1–9.
- [144] N. Abbani, M. Jomaa, T. Tarhini, H. Artail, and W. El-Hajj, "Managing social networks in vehicular networks using trust rules," in *Proc. Symp. Wireless Technol. Appl. (ISWTA)*, Sep. 2011, pp. 168–173.
- [145] Z. Huang, S. Ruj, M. A. Cavenaghi, M. Stojmenovic, and A. Nayak, "A social network approach to trust management in VANETs," *Peer Peer Netw. Appl.*, vol. 7, no. 3, pp. 229–242, 2014.
- [146] X. Chen and L. Wang, "A cloud-based trust management framework for vehicular social networks," *IEEE Access*, vol. 5, pp. 2967–2980, 2017.
- [147] T. R. de Oliveira, S. de Oliveira, D. F. Macedo, and J. M. Nogueira, "Social networks for certification in vehicular disruption tolerant networks," in *Proc. Int. Conf. Wireless Mobile Comput., Netw. Commun.* (*WiMob*), 2014, pp. 479–486.
- [148] A. Srivastava, Anuradha, and D. J. Gupta, "Social network analysis: Hardly easy," in *Proc. Int. Conf. Optim., Reliability, Inf. Technol.* (*ICROIT*), 2014, pp. 128–135.
- [149] A. Bradai and T. Ahmed, "ReViV: Selective rebroadcast mechanism for video streaming over VANET," in *Proc. Veh. Technol. Conf. (VTC Spring)*, May 2014, pp. 1–6.
- [150] F. D. Cunha, G. G. Maia, A. C. Viana, R. A. Mini, L. A. Villas, and A. A. Loureiro, "Socially inspired data dissemination for vehicular ad hoc networks," in *Proc. 17th ACM Int. Conf. Modeling Anal. Simulation Wireless Mobile Syst.*, 2014, pp. 81–85.
- [151] A. Stagkopoulou, P. Basaras, and D. Katsaros, "A social-based approach for message dissemination in vehicular ad hoc networks," in *Proc. Int. Conf. Ad Hoc Netw.*. Benidorm, Spain: Springer, 2014, pp. 27–38.
- [152] L. A. Maglaras and D. Katsaros, "Social clustering of vehicles based on semi-Markov processes," *IEEE Trans. Veh. Technol.*, vol. 65, no. 1, pp. 318–332, Jan. 2016.
- [153] S. Vodopivec, J. Bešter, and A. Kos, "A survey on clustering algorithms for vehicular ad-hoc networks," in *Proc. Int. Conf. Telecommun. Signal Process. (TSP)*, 2012, pp. 52–56.

- [154] X. Hu, J. Zhao, B. C. Seet, V. C. M. Leung, T. H. S. Chu, and H. Chan, "S-Aframe: Agent-based multilayer framework with context-aware semantic service for vehicular social networks," *IEEE Trans. Emerg. Topics Comput.*, vol. 3, no. 1, pp. 44–63, Mar. 2015.
- [155] A.-U.-H. Yasar, N. Mahmud, D. Preuveneers, K. Luyten, K. Coninx, and Y. Berbers, "Where people and cars meet: Social interactions to improve information sharing in large scale vehicular networks," in *Proc. ACM Symp. Appl. Comput.*, 2010, pp. 1188–1194.
- [156] R. Lu, X. Lin, X. Liang, and X. Shen, "Sacrificing the plum tree for the peach tree: A socialspot tactic for protecting receiver-location privacy in VANET," in *Proc. IEEE Global Telecommun. Conf. (GLOBECOM)*, Dec. 2010, pp. 1–5.
- [157] A. Alganas, X. Lin, and A. Grami, "EVSE: An efficient vehicle social evaluation scheme with location privacy preservation for vehicular communications," in *Proc. Int. Conf. Commun. (ICC)*, 2011, pp. 1–5.
- [158] X. Hu *et al.*, "Social drive: A crowdsourcing-based vehicular social networking system for green transportation," in *Proc. 3rd ACM Int. Symp. Design Anal. Intell. Veh. Netw. Appl.*, 2013, pp. 85–92.
- [159] H. Gong, L. Yu, and X. Zhang, "Social contribution-based routing protocol for vehicular network with selfish nodes," *Int. J. Distrib. Sensor Netw.*, vol. 10, no. 4, p. 753024, 2014.
- [160] F. Xia, L. Liu, J. Li, A. M. Ahmed, L. T. Yang, and J. Ma, "BEEINFO: Interest-based forwarding using artificial bee colony for socially aware networking," *IEEE Trans. Veh. Technol.*, vol. 64, no. 3, pp. 1188–1200, Mar. 2015.
- [161] R. L. Villars, C. W. Olofson, and M. Eastwood, "Big data: What it is and why you should care," IDC, Framingham, MA, USA, White Paper, 2011, vol. 14, pp. 1–14.
- [162] C.-W. Lu, C.-M. Hsieh, C.-H. Chang, and C.-T. Yang, "An improvement to data service in cloud computing with content sensitive transaction analysis and adaptation," in *Proc. IEEE 37th Annu. Comput. Softw. Appl. Conf. Workshops (COMPSACW)*, Jul. 2013, pp. 463–468.
- [163] W. He, G. Yan, and L. D. Xu, "Developing vehicular data cloud services in the IoT environment," *IEEE Trans. Ind. Inform.*, vol. 10, no. 2, pp. 1587–1595, May 2014.
- [164] D. J. DeWitt, S. Ghandeharizadeh, D. A. Schneider, A. Bricker, H.-I. Hsiao, and R. Rasmussen, "The Gamma database machine project," *IEEE Trans. Knowl. Data Eng.*, vol. 2, no. 1, pp. 44–62, Mar. 1990.
- [165] D. Agrawal, S. Das, and A. El Abbadi, "Big data and cloud computing: Current state and future opportunities," in *Proc. 14th Int. Conf. Extending Database Technol.*, 2011, pp. 530–533.
- [166] K. Chard, K. Bubendorfer, S. Caton, and O. F. Rana, "Social cloud computing: A vision for socially motivated resource sharing," *IEEE Trans. Services Comput.*, vol. 5, no. 4, pp. 551–563, Jun. 2012.
- [167] X. Wang, L. T. Wang, X. Xie, J. Jin, and M. J. Deen, "A cloud-edge computing framework for cyber-physical-social services," *IEEE Commun. Mag.*, vol. 55, no. 11, pp. 80–85, Nov. 2017.
- [168] D. Ardagna, M. Ciavotta, and M. Passacantando, "Generalized nash equilibria for the service provisioning problem in multi-cloud systems," *IEEE Trans. Services Comput.*, vol. 10, no. 3, pp. 381–395, May/Jun. 2017.
- [169] Y. Wang, C. Guo, T. Li, and Q. Xu, "Notice of violation of ieee publication principles secure two-party computation in social cloud based on reputation," in *Proc. Int. Conf. Adv. Inf. Netw. Appl. Workshops (WAINA)*, 2015, pp. 242–245.
- [170] M. Z. Bidoki and M. J. Kargar, "A social cloud computing: Employing a bee colony algorithm for sharing and allocating tourism resources," *Mod. Appl. Sci.*, vol. 10, no. 5, p. 177, 2016.
- [171] S. Caton, C. Haas, K. Chard, K. Bubendorfer, and O. F. Rana, "A social compute cloud: Allocating and sharing infrastructure resources via social networks," *IEEE Trans. Services Comput.*, vol. 7, no. 3, pp. 359–372, Jul./Sep. 2014.
- [172] K. Chard and S. Caton, "Social clouds: Crowdsourcing cloud infrastructure," in *Crowdsourcing*. Berlin, Germany: Springer-Verlag, 2015, pp. 191–217.
- [173] H. Jagadish et al., "Big data and its technical challenges," Commun. ACM, vol. 57, no. 7, pp. 86–94, 2014.
- [174] M. Zaharia et al., "Apache spark: A unified engine for big data processing," Commun. ACM, vol. 59, no. 11, pp. 56–65, 2016.
- [175] J. Quevedo, D. Corujo, and R. Aguiar, "A case for ICN usage in IoT environments," in *Proc. IEEE Global Commun. Conf. (GLOBECOM)*, Dec. 2014, pp. 2770–2775.
- [176] M. A. Yaqub, S. H. Ahmed, S. H. Bouk, and D. Kim, "Informationcentric networks (ICN)," in *Content-Centric Networks*. Berlin, Germany: Springer-Verlag, 2016, pp. 19–33.

- [177] H. Luo, Z. Chen, J. Cui, H. Zhang, M. Zukerman, and C. Qiao, "CoLoR: An information-centric Internet architecture for innovations," *IEEE Netw.*, vol. 28, no. 3, pp. 4–10, May/Jun. 2014.
- [178] L. Zhang et al., "Named data networking," SIGCOMM Comput. Commun. Rev., vol. 44, no. 3, pp. 66–73, Jul. 2014.
- [179] A. Friggeri, G. Chelius, and E. Fleury, "Ego-munities, Exploring Socially Cohesive Person-based Communities," Inria, Tech. Rep. RR-7535, 2011.
- [180] M. Franceschet, "PageRank: Standing on the shoulders of giants," Commun. ACM, vol. 54, no. 6, pp. 92–101, Jun. 2011.
- [181] A. N. Langville and C. D. Meyer, "A survey of eigenvector methods for Web information retrieval," SIAM Rev., vol. 47, no. 1, pp. 135–161, 2005.
- [182] C. Bernardini, T. Silverston, and O. Festor, "Socially-aware caching strategy for content centric networking," in *Proc. IFIP Netw. Conf.*, 2014, pp. 1–9.
- [183] B. Mathieu, P. Truong, W. You, and J.-F. Peltier, "Information-centric networking: A natural design for social network applications," *IEEE Commun. Mag.*, vol. 50, no. 7, pp. 44–51, Jul. 2012.
- [184] W. Tan, M. B. Blake, I. Saleh, and S. Dustdar, "Social-network-sourced big data analytics," *IEEE Internet Comput.*, vol. 17, no. 5, pp. 62–69, Sep. 2013.
- [185] C. Gao and N. Iwane, "A social network model for big data privacy preserving and accountability assurance," in *Proc. Consum. Commun. Netw. Conf. (CCNC)*, Jan. 2015, pp. 19–22.
- [186] O. Johny, S. Sotiriadis, E. Asimakopoulou, and N. Bessis, "Towards a social graph approach for modeling risks in big data and Internet of Things (IoT)," in *Proc. Int. Conf. Intell. Netw. Collaborative Syst.* (INCoS), 2014, pp. 439–444.
- [187] G. Bello-Orgaz, J. J. Jung, and D. Camacho, "Social big data: Recent achievements and new challenges," *Inf. Fusion*, vol. 28, pp. 45–59, Mar. 2016.
- [188] C. Zhu, H. Wang, V. C. Leung, L. Shu, and L. T. Yang, "An evaluation of user importance when integrating social networks and mobile cloud computing," in *Proc. Global Commun. Conf. (GLOBECOM)*, 2014, pp. 2935–2940.
- [189] Z. Li and M. Li, "An efficient social based data forwarding mechanism for mobile cloud computing," in *Proc. Int. Conf. Mobile Ad-Hoc Sensor Netw. (MSN)*, 2013, pp. 365–372.
- [190] Y. Wu, Z. Zhang, C. Wu, Z. Li, and F. Lau, "CloudMoV: Cloud-based mobile social TV," *IEEE Trans. Multimedia*, vol. 15, no. 4, pp. 821–832, Jun. 2012.
- [191] X. Wang, T. Kwon, Y. Choi, H. Wang, and J. Liu, "Cloud-assisted adaptive video streaming and social-aware video prefetching for mobile users," *IEEE Wireless Commun.*, vol. 20, no. 3, pp. 72–79, Jun. 2013.
- [192] Q. Xu, Z. Su, S. Yu, and Y. Wang, "Trust based incentive scheme to allocate big data tasks with mobile social cloud," *IEEE Trans. Big Data*, to be published.
- [193] H. Hu, Y. Wen, and D. Niyato, "Public cloud storage-assisted mobile social video sharing: A supermodular game approach," *IEEE J. Sel. Areas Commun.*, vol. 35, no. 3, pp. 545–556, Mar. 2017.
- [194] I. Petri, J. Diaz-Montes, O. Rana, M. Punceva, I. Rodero, and M. Parashar, "Modelling and implementing social community clouds," *IEEE Trans. Services Comput.*, vol. 10, no. 3, pp. 410–422, May 2017.
- [195] S. Agarwal and S. Agarwal, "Social networks as Internet barometers for optimizing content delivery networks," in *Proc. Int. Symp. Adv. Netw. Telecommun. Syst. (ANTS)*, 2009, pp. 1–3.
- [196] S. Scellato, C. Mascolo, M. Musolesi, and J. Crowcroft, "Track globally, deliver locally: Improving content delivery networks by tracking geographic social cascades," in *Proc. 20th Int. Conf. World Wide Web*, 2011, pp. 457–466.
- [197] F. Nazir, J. Ma, and A. Seneviratne, "Time critical content delivery using predictable patterns in mobile social networks," in *Proc. Int. Conf. Comput. Sci. Eng.*, Aug. 2009, pp. 1066–1073.
- [198] C. Fang, F. R. Yu, T. Huang, J. Liu, and Y. Liu, "Energy-efficient distributed in-network caching for content-centric networks," in *Proc. Conf. Comput. Commun. Workshops (INFOCOM WKSHPS)*, 2014, pp. 91–96.
- [199] J. Kim, M.-W. Jang, B.-J. Lee, and K. Kim, "Content centric networkbased virtual private community," in *Proc. Int. Conf. Consum. Electron.* (*ICCE*), 2011, pp. 843–844.
- [200] D. Kim and J. Lee, "CCN-based virtual private community for extended home media service," *IEEE Trans. Consum. Electron.*, vol. 57, no. 2, pp. 532–540, May 2011.

- [201] R. G. Garroppo, G. Nencioni, G. Procissi, and L. Tavanti, "The energy footprint of content-centric residential community networks," in *Proc. Tyrrhenian Int. Workshop Digit. Commun.-Green ICT (TIWDC)*, 2013, pp. 1–6.
- [202] M. Ammar, G. Russello, and B. Crispo, "Internet of Things: A survey on the security of IoT frameworks," *J. Inf. Secur. Appl.*, vol. 38, pp. 8–27, Feb. 2018.
- [203] S. Agnihotri, S. Agnihotri, K. Ramkumar, and K. Ramkumar, "A survey and comparative analysis of the various routing protocols of Internet of Things," *Int. J. Pervasive Comput. Commun.*, vol. 13, no. 3, pp. 264–281, 2017.
- [204] J. D. Lewis and A. Weigert, "Trust as a social reality," *Social forces*, vol. 63, no. 4, pp. 967–985, 1985.
- [205] F. D. Schoorman, R. C. Mayer, and J. H. Davis, "An integrative model of organizational trust: Past, present, and future," *Acad. Manage. Rev.*, vol. 32, no. 2, pp. 344–354, 2007.
- [206] W. Z. Khan, M. Y. Aalsalem, M. K. Khan, and Q. Arshad, "When social objects collaborate: Concepts, processing elements, attacks and challenges," *Comput. Elect. Eng.*, vol. 58, pp. 397–411, Feb. 2017.
- [207] J. F. Colom, H. Mora, D. Gil, and M. T. Signes-Pont, "Collaborative building of behavioural models based on Internet of Things," *Comput. Elect. Eng.*, vol. 58, pp. 385–396, Feb. 2017.
- [208] P. Kasnesis, C. Z. Patrikakis, D. Kogias, L. Toumanidis, and I. S. Venieris, "Cognitive friendship and goal management for the social IoT," *Comput. Elect. Eng.*, vol. 58, pp. 412–428, Feb. 2017.
- [209] D. Hussein, S. N. Han, G. M. Lee, N. Crespi, and E. Bertin, "Towards a dynamic discovery of smart services in the social Internet of Things," *Comput. Elect. Eng.*, vol. 58, pp. 429–443, Feb. 2017.
- [210] P. N. Mahalle, P. A. Thakre, N. R. Prasad, and R. Prasad, "A fuzzy approach to trust based access control in Internet of Things," in *Proc. Int. Conf. Wireless Commun., Veh. Technol., Inf. Theory Aerosp. Electron. Syst. (VITAE)*, Jun. 2013, pp. 1–5.
- [211] S. Brin and L. Page, "Reprint of: The anatomy of a large-scale hypertextual Web search engine," *Comput. Netw.*, vol. 56, no. 18, pp. 3825–3833, 2012.
- [212] U. Jayasinghe, N. B. Truong, G. M. Lee, and T.-W. Um, "RpR: A trust computation model for social Internet of Things," in *Proc. Ubiquitous Intell. Comput., Adv. Trusted Comput., Scalable Comput. Commun., Cloud Big Data Comput., Internet People, Smart World Congr. (UIC/ATC/ScalCom/CBDCom/IoP/SmartWorld), Jul. 2016, pp. 930–937.*
- [213] J. B. Bernabe, J. L. H. Ramos, and A. F. S. Gomez, "TACIoT: Multidimensional trust-aware access control system for the Internet of Things," *Soft Comput.*, vol. 20, no. 5, pp. 1763–1779, 2016.
- [214] S. E. A. Rafey, A. Abdel-Hamid, and M. A. El-Nasr, "CBSTM-IoT: Context-based social trust model for the Internet of Things," in *Proc. Int. Conf. Sel. Topics Mobile Wireless Netw. (MoWNeT)*, Apr. 2016, pp. 1–8.
- [215] M. Nitti, L. Atzori, and I. P. Cvijikj, "Friendship selection in the social Internet of Things: Challenges and possible strategies," *IEEE Internet Things J.*, vol. 2, no. 3, pp. 240–247, Jun. 2015.
- [216] J. An, X. Gui, W. Zhang, J. Jiang, and J. Yang, "Research on social relations cognitive model of mobile nodes in Internet of Things," J. Netw. Comput. Appl., vol. 36, no. 2, pp. 799–810, 2013.
- [217] L. Atzori, D. Carboni, and A. Iera, "Smart things in the social loop: Paradigms, technologies, and potentials," *Ad Hoc Netw.*, vol. 18, pp. 121–132, Jul. 2014.
- [218] A. Zelenkauskaite, N. Bessis, S. Sotiriadis, and E. Asimakopoulou, "Interconnectedness of complex systems of Internet of Things through social network analysis for disaster management," in *Proc. Int. Conf. Intell. Netw. Collaborative Syst. (INCoS)*, Sep. 2012, pp. 503–508.
- [219] T. H. Luan, R. Lu, X. Shen, and F. Bai, "Social on the road: Enabling secure and efficient social networking on highways," *IEEE Wireless Commun.*, vol. 22, no. 1, pp. 44–51, Feb. 2015.
- [220] K. M. Alam, M. Saini, and A. E. Saddik, "Toward social Internet of vehicles: Concept, architecture, and applications," *IEEE Access*, vol. 3, pp. 343–357, Mar. 2015.
- [221] V. Sharma, I. You, D. N. K. Jayakody, and M. Atiquzzaman, "Cooperative trust relaying and privacy preservation via edge-crowdsourcing in social Internet of Things," *Future Gener. Comput. Syst.*, to be published.
- [222] S. Ahmed and S. S. Kanhere, "A Bayesian routing framework for delay tolerant networks," in *Proc. Wireless Commun. Netw. Conf. (WCNC)*, 2010, pp. 1–6.

- [223] R. Gupta, N. Krishnamurthi, U.-T. Wang, T. Tamminedi, and M. Gerla, "Routing in mobile Ad-Hoc networks using social tie strengths and mobility plans," in *Proc. IEEE Wireless Commun. Netw. Conf. (WCNC)*, Mar. 2017, pp. 1–6.
- [224] Z. Ding, F. Adachi, and H. V. Poor, "The application of MIMO to nonorthogonal multiple access," *IEEE Trans. Wireless Commun.*, vol. 15, no. 1, pp. 537–552, Jan. 2016.
- [225] S. Qureshi, S. A. Hassan, and D. N. K. Jayakody, "Divide-and-allocate: An uplink successive bandwidth division NOMA system," *Trans. Emerg. Telecommun. Technol.*, vol. 29, no. 1, p. e3216, 2018.
- [226] Z. Ding et al., "Application of non-orthogonal multiple access in LTE and 5G networks," *IEEE Commun. Mag.*, vol. 55, no. 2, pp. 185–191, Feb. 2017.
- [227] Z. Ding, M. Peng, and H. V. Poor, "Cooperative non-orthogonal multiple access in 5G systems," *IEEE Commun. Lett.*, vol. 19, no. 8, pp. 1462–1465, Aug. 2015.
- [228] W. Shin, M. Vaezi, B. Lee, D. J. Love, J. Lee, and H. V. Poor, "Non-orthogonal multiple access in multi-cell networks: Theory, performance, and practical challenges," *IEEE Commun. Mag.*, vol. 55, no. 10, pp. 176–183, Oct. 2017.
- [229] S. Qureshi, S. A. Hassan, and D. N. K. Jayakody, "Successive bandwidth division noma systems: Uplink power allocation with proportional fairness," in *Proc. 14th IEEE Annu. Consum. Commun. Netw. Conf. (CCNC)*, 2017, pp. 998–1003.
- [230] A. V. N. Jalajakumari *et al.*, "High-speed integrated digital to light converter for short range visible light communication," *IEEE Photon. Technol. Lett.*, vol. 29, no. 1, pp. 118–121, Jan. 1, 2017.
- [231] S. Chedup *et al.*, "Visible light energy harvesting in modern communication systems," in *Proc. EECCMC*, 2015, pp. 252–257.
- [232] G. Corbellini, K. Aksit, S. Schmid, S. Mangold, and T. Gross, "Connecting networks of toys and smartphones with visible light communication," *IEEE Commun. Mag.*, vol. 52, no. 7, pp. 72–78, Jul. 2014.
- [233] S. Li, A. Pandharipande, and F. M. J. Willems, "Two-way visible light communication and illumination with LEDs," *IEEE Trans. Commun.*, vol. 65, no. 2, pp. 740–750, Feb. 2017.
- [234] H. Haas, L. Yin, Y. Wang, and C. Chen, "What is LiFi?" J. Lightw. Technol., vol. 34, no. 6, pp. 1533–1544, Mar. 15, 2016.
- [235] A. Mostafa and L. Lampe, "Physical-layer security for MISO visible light communication channels," *IEEE J. Sel. Areas Commun.*, vol. 33, no. 9, pp. 1806–1818, Sep. 2015.
- [236] N. Lourenço, D. Terra, N. Kumar, L. N. Alves, and R. L. Aguiar, "Visible light communication system for outdoor applications," in *Proc. Int. Symp. Commun. Syst., Netw. Digit. Signal Process. (CSNDSP)*, 2012, pp. 1–6.
- [237] O. A. Sadik, S. K. Mwilu, and A. Aluoch, "Smart electrochemical biosensors: From advanced materials to ultrasensitive devices," *Electrochimica Acta*, vol. 55, no. 14, pp. 4287–4295, May 2010.
- [238] K. Aziz, S. Tarapiah, S. H. Ismail, and S. Atalla, "Smart real-time healthcare monitoring and tracking system using GSM/GPS technologies," in *Proc. MEC Int. Conf. Big Data Smart City (ICBDSC)*, 2016, pp. 1–7.
- [239] S. Sendra, L. Parra, J. Lloret, and J. Tomás, "Smart system for children's chronic illness monitoring," *Inf. Fusion*, vol. 40, pp. 76–86, Mar. 2018.
- [240] M. Amadeo, A. Molinaro, and G. Ruggeri, "An energy-efficient contentcentric approach in mesh networking," in *Proc. Int. Conf. Commun.* (*ICC*), Jun. 2012, pp. 5736–5740.
- [241] S. Y. Oh, D. Lau, and M. Gerla, "Content centric networking in tactical and emergency manets," in *Proc. IFIP Wireless Days (WD)*, 2010, pp. 1–5.
- [242] Y. Shim and Y. Kim, "Data aggregation with multiple sinks in information-centric wireless sensor network," in *Proc. Int. Conf. Inf. Netw. (ICOIN)*, 2014, pp. 13–17.
- [243] R.-I. Chang and C.-C. Chuang, "A service-oriented cloud computing network management architecture for wireless sensor networks," *Ad-Hoc Sensor Wireless Netw.*, vol. 22, nos. 1–2, pp. 65–90, 2014.
- [244] Y. Guo, F. Hong, and Z. Guo, "Event-oriented data aggregation in wireless sensor networks," *Ad Hoc Sensor Wireless Netw.*, vol. 23, nos. 3–4, pp. 297–327, 2014.
- [245] M.-S. Jin, H. Park, E. Lee, S. Park, and S.-H. Kim, "A data dissemination model base on content-based publish/subscribe paradigm in large-scale wireless sensor networks," in *Proc. Wireless Commun. Netw. Conf. (WCNC)*, Apr. 2009, pp. 1–6.
- [246] C. Liang, F. R. Yu, H. Yao, and Z. Han, "Virtual resource allocation in information-centric wireless networks with virtualization," *IEEE Trans. Veh. Technol.*, vol. 65, no. 12, pp. 9902–9914, Dec. 2016.

- [247] C. Liang and F. R. Yu, "Virtual resource allocation in information-centric wireless virtual networks," in *Proc. Int. Conf. Commun. (ICC)*, Jun. 2015, pp. 3915–3920.
- [248] A. Aktas, K. Erhan, S. Ozdemir, and E. Ozdemir, "Experimental investigation of a new smart energy management algorithm for a hybrid energy storage system in smart grid applications," *Electr. Power Syst. Res.*, vol. 144, pp. 185–196, Mar. 2017.
- [249] F. Skopik, "The social smart grid: Dealing with constrained energy resources through social coordination," J. Syst. Softw., vol. 89, pp. 3–18, Mar. 2014.
- [250] I. G. Ciuciu, R. Meersman, and T. Dillon, "Social network of smartmetered homes and smes for grid-based renewable energy exchange," in *Proc. Int. Conf. Digit. Ecosyst. Technol. (DEST)*, 2012, pp. 1–6.
- [251] A. Cassidy, M. Strube, and A. Nehorai, "A framework for exploring social network and personality-based predictors of smart grid diffusion," *IEEE Trans. Smart Grid*, vol. 6, no. 3, pp. 1314–1322, May 2015.
- [252] Q. Huang, X. Li, J. Zhao, D. Wu, and X.-Y. Li, "Social networking reduces peak power consumption in smart grid," *IEEE Trans. Smart Grid*, vol. 6, no. 3, pp. 1403–1413, May 2015.
- [253] N. Good, K. A. Ellis, and P. Mancarella, "Review and classification of barriers and enablers of demand response in the smart grid," *Renew. Sustain. Energy Rev.*, vol. 72, pp. 57–72, May 2017.
- [254] J. C. Tsai, N. Y. Yen, and T. Hayashi, "Social network based smart grids analysis," in *Proc. Int. Symp. Independ. Comput. (ISIC)*, Dec. 2014, pp. 1–6.
- [255] A. Cassidy and A. Nehorai, "Modeling smart grid adoption via a social network model," in Proc. PES Gen. Meeting, Conf. Expo., 2014, pp. 1–5.
- [256] J. G. Andrews et al., "What will 5G be?" IEEE J. Sel. Areas Commun., vol. 32, no. 6, pp. 1065–1082, 2014.
- [257] N. S. Perović, P. Liu, M. Di Renzo, and A. Springer, "Receive spatial modulation for LOS mmWave communications based on TX beamforming," *IEEE Commun. Lett.*, vol. 21, no. 4, pp. 921–924, Apr. 2017.
- [258] Z. Huang, H. Tian, C. Qin, S. Fan, and X. Zhang, "A social-energy based cluster management scheme for user-centric ultra-dense networks," *IEEE Access*, vol. 5, pp. 10769–10781, 2017.
- [259] A. Beylerian and T. Ohtsuki, "Service-aware user-centric clustering and scheduling for cloud-ran with coordinated multi-point transmission," in *Proc. Signal Inf. Process. Assoc. Annu. Summit Conf. (APSIPA)*, 2015, pp. 252–257.
- [260] M. I. Ashraf, M. Bennis, W. Saad, M. Katz, and C.-S. Hong, "Dynamic clustering and user association in wireless small-cell networks with social considerations," *IEEE Trans. Veh. Technol.*, vol. 66, no. 7, pp. 6553–6568, Jul. 2017.



FURQAN JAMEEL received the B.S. degree in electrical engineering (under ICT Research and Development Funded Program) from the COM-SATS Institute of Information Technology (CIIT), Lahore Campus, Pakistan, in 2013, and the master's degree in electrical engineering (funded by prestigious Higher Education Commission Scholarship) from CIIT, Islamabad Campus, in 2017. His research interests include modeling and performance enhancement of vehicular networks,

physical layer security, ambient backscatter communications, and wireless power transfer. He received the Outstanding Reviewer Award from Elsevier in 2017.



SHURJEEL WYNE (S'02–M'08–SM'13) received the Ph.D. degree from Lund University, Sweden, in 2009. Between 2009 and 2010, he was a Post -Doctoral Research Fellow with Lund University, funded by the High-Speed Wireless Center. Since 2010, he has been with the Department of Electrical Engineering, COMSATS University Islamabad, Pakistan, where he currently serves as an Associate Professor. His research interests include wireless channel characterization and modeling,

cooperative communications, multi-antenna systems, physical layer security, and vehicular communications. He was a co-recipient of the Best Paper Award of the Antennas and Propagation Track in IEEE VTC2013-Spring.



DUSHANTHA NALIN K. JAYAKODY (S'09– M'14–SM'18) received the B.E. degree (Hons.) in electronics engineering from the Dawood University of Engineering and Technology, Karachi, Pakistan (under SAARC Scholarship), in 2008, the M.Sc. degree in electronics and communications engineering from the Department of Electrical and Electronics Engineering, Eastern Mediterranean University, Turkey (under the University Full Graduate Scholarship), in 2009, and

the Ph.D. degree in electronics, electrical, and communications engineering from University College Dublin, Ireland, in 2014. From 2014 to 2016, he was a Post-Doctoral Research Fellow with the Institute of Computer Science, University of Tartu, Estonia, and also with the Department of Informatics, University of Bergen, Norway. Since 2016, he has been a Professor with the School of Computer Science and Robotics, National Research Tomsk Polytechnic University, Russia, where he is also the Director of the Tomsk Infocomm Lab. He has authored over 80 international peer-reviewed journal and conference papers. His research interests include PHY layer prospective of 5G communications, cooperative wireless communications, device-todevice communications, LDPC codes, and unmanned aerial vehicle. He has served as the Workshop Chair, the Session Chair, and a Technical Program Committee Member for various international conferences, such as IEEE PIMRC from 2013 to 2014, IEEE WCNC from 2014 to 2018, and IEEE VTC from 2015 to 2018. He received the Best Paper Award from the IEEE International Conference on Communication, Management and Information Technology in 2017. He currently serves as an Area Editor for the Physical Communications Journal (Elsevier), Information Journal (MDPI), and Internet of Technology Letters (Wiley). He serves as a reviewer for various IEEE transactions and other journals.



GEORGES KADDOUM (M'11) received the bachelor's degree in electrical engineering from the Ecole Nationale Superieure de Techniques Avancees de Bretagne, Brest, France, the M.S. degree in telecommunications and signal processing (circuits, systems, and signal processing) from the Universite de Bretagne Occidentale and Telecom Bretagne, Brest, in 2005, and the Ph.D. degree (Hons.) in signal processing and telecommunications from the National Institute of Applied Sci-

ences, University of Toulouse, Toulouse, France, in 2009. Since 2013, he has been an Assistant Professor of electrical engineering with ETS, University of Quebec, Montreal, QC, Canada. He has authored over 150+ journal and conference papers and holds two pending patents. His recent research activities cover mobile communication systems, modulations, secure transmissions, and space communication systems, modulations, secure transmissions, and space communication and navigation. He was awarded as the ETS Research Chair of physical-layer security for wireless networks in 2014. He was a recipient of the 2015 and 2017 IEEE Transactions on Communications Exemplary Reviewer Award. He received the best papers awards at the 2014 IEEE International Conference on Wireless and Mobile Computing, Networking, Communications, with three co-authors, and at the 2017 IEEE International Symposium on Personal Indoor and Mobile Radio Communications, with four co-authors. He currently serves as an Associate Editor for the IEEE TRANSACTIONS ON INFORMATION FORENSICS AND SECURITY and IEEE COMMUNICATIONS LETTERS.



RICHARD O'KENNEDY received the B.Sc. and Ph.D. degrees from University College Dublin, the Dip. F.S. degree from the Dublin Institute of Technology, and the Dip. C.S. and Dip. Computing degrees from Trinity College Dublin. He was a Vice-President of research with Hamad Bin Khalifa University, Doha, Qatar, and a Professor and the Scientific Director of the Biomedical Diagnostics Institute, Dublin City University (DCU). He was a Vice-President of teaching and learning with DCU

and the Visiting Scientist with the M.D. Anderson Cancer Centre, The University of Texas at Houston, where his work was selected as Invention of the Year. He is a Vice-President of research, development and innovation with Qatar Foundation. He directs the Applied Biochemistry Research Group, internationally recognized for its expertise in antibody generation and immuno/biosensor assay development. He has authored extensively (250 peer-reviewed papers, 51 reviews, 40 book chapters, 2 books, and 1 in process) and mentored 69 Ph.D. students to completion. He has a Google Scholar *h*-index of 52 with ca 11 600 citations. He has academic collaborations worldwide and carried out many collaborations with industry. He holds seven patents and multiple licenses, and many of his reagents/innovations have been licensed and commercialized. He has a major interest in innovative

approaches to healthcare. He is member of Royal Irish Academy (RIA) and a fellow of Royal Society of Biology. He was a member of the Industrial Research and Commercialisation Committee of Enterprise Ireland, the National Bioethics Council for Ireland, and the Life Sciences and Health Committee of RIA. He is a fellow of the Institute of Biology of Ireland. He received many awards for research (Biochemistry Medal of the RIA, President's Award for Research, three Fujitsu awards for Innovation, and Ireland's Bioscience Lab of the Year in 2014) and teaching (President's Award from DCU, Outstanding Masters Programme Award, Graduate Ireland in 2011, and Outstanding Education and Outreach Lab, 2017). He was the Chair of the Management Committee of the Dublin Region Higher Education Alliance (11 institutions and ca 66 000 students). He was the President of the London International Youth Science Forum and the Institute of Biology of Ireland. He was the Chairman/Founder of the Centre for Talented Youth, Ireland, and represents Ireland and RIA on the International Union of Biochemistry and Molecular Biology. He is a Founder and the Chief Scientific Officer of AbYBiotech, and the Chairman of the Scientific Board of Glycoselect, a highly innovative Biotech company. He is a Chartered Biologist from the Royal Society of Biology, U.K. He has acted as a reviewer for over 60 journals and for ca 30 international funding bodies. He is an Editor of three journals.

...